

LUND UNIVERSITY

Testicular cancer; gonadal, sexual and psychological aspects of the disease and its treatment.

Eberhard, Jakob

2009

Link to publication

Citation for published version (APA):

Eberhard, J. (2009). *Testicular cancer; gonadal, sexual and psychological aspects of the disease and its treatment.* [Doctoral Thesis (compilation), Breastcancer-genetics]. Jakob Eberhard, Dep of Clinical Sciences, Malmö University Hospital, Lund University.

Total number of authors: 1

General rights

Unless other specific re-use rights are stated the following general rights apply: Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights. • Users may download and print one copy of any publication from the public portal for the purpose of private study

or research.

- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal

Read more about Creative commons licenses: https://creativecommons.org/licenses/

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

LUND UNIVERSITY

PO Box 117 221 00 Lund +46 46-222 00 00

and attendes

Testicular cancer; gonadal, sexual and psychological aspects of the disease and its treatment

(cool

Geed!

Jakob Eberhard

Testicular cancer: gonadal, sexual and psychological aspects of the disease and its treatment.

Jakob Eberhard Department of Clinical Sciences Molecular Reproductive Medicine Research Unit Malmö University Hospital



LUND UNIVERSITY Faculty of Medicine

Academic Dissertation

with permission of the Medical Faculty of Lund University to be presented for public defence in Kvinnoklinikens aula, entrance 74, floor 3, Malmö University Hospital

Friday, April 17th, 2009 at 09:00 am

Faculty Opponent: Professor Mikael Rørth, Department of Oncology, Rigshospitalet, Copenhagen

Testicular cancer: gonadal, sexual and psychological aspects of the disease and its treatment.

Jakob Eberhard Department of Clinical Sciences Molecular Reproductive Medicine Research Unit Malmö University Hospital



LUND UNIVERSITY Faculty of Medicine

© 2009 Jakob Eberhard ISSN 1652-8220 ISBN 978-91-86253-18-9 Cover page: Maria Collinder, www.renform.com Lund University, Faculty of Medicine Doctoral Dissertation Series 2009:31 Printed by Media-Tryck, Lund 2009

Table of Contents

Table of Contents	5
Abbreviations	7
List of Original Papers	
Popular scientific summary	0 0
Rockaround	13
Tackgi vullar aanaar	12
Findemiology	13
Aetiology	13
Diagnosis and staging	13
Treatment	14
Spermatogenesis and male endocrinology	17
Testicular histology	17
The hypothalamic pituitary testicular axis	
Androgen production and function	
Spermatogenesis	
Infertility	
Testicular cancer. side-effects of disease and its treatment	
Testicular cancer and infertility	
Testicular cancer and hypogonadism	
Testicular cancer and sexual dysfunction	
Testicular cancer and emotional disorders	35
Aims of the Thesis	
Aims of the Thesis Material and Methods	
Aims of the Thesis Material and Methods Patient inclusion	
Aims of the Thesis Material and Methods Patient inclusion Methods	
Aims of the Thesis Material and Methods Patient inclusion Methods Cancer treatment (articles I-IV)	
Aims of the Thesis Material and Methods Patient inclusion Methods Cancer treatment (articles I-IV) Sperm (article I) and hormone analyses (articles II-IV).	
Aims of the Thesis Material and Methods Patient inclusion Methods Cancer treatment (articles I-IV) Sperm (article I) and hormone analyses (articles II-IV) DNA analysis (articles I. II and IV)	39 41 43 43 43 43
Aims of the Thesis Material and Methods Patient inclusion Methods Cancer treatment (articles I-IV) Sperm (article I) and hormone analyses (articles II-IV) DNA analysis (articles I, II and IV) Evaluation of sexuality and socio-demographics (article III)	39 41 43 43 43 43 43 45 45
Aims of the Thesis Material and Methods Patient inclusion Methods Cancer treatment (articles I-IV) Sperm (article I) and hormone analyses (articles II-IV) DNA analysis (articles I, II and IV) Evaluation of sexuality and socio-demographics (article III) Measures of emotional disorders (article IV)	39 41 43 43 43 43 43 43 45 45 46
Aims of the Thesis Material and Methods Patient inclusion Methods Cancer treatment (articles I-IV) Sperm (article I) and hormone analyses (articles II-IV) DNA analysis (articles I, II and IV) Evaluation of sexuality and socio-demographics (article III) Measures of emotional disorders (article IV) Testicular characteristics (article II)	39 41 43 43 43 43 45 45 46 47
Aims of the Thesis Material and Methods Patient inclusion Methods Cancer treatment (articles I-IV) Sperm (article I) and hormone analyses (articles II-IV) DNA analysis (articles I, II and IV) Evaluation of sexuality and socio-demographics (article III) Measures of emotional disorders (article IV) Testicular characteristics (article II) Statistical analysis	39 41 43 43 43 43 45 45 45 46 47 47
Aims of the Thesis Material and Methods Patient inclusion Methods Cancer treatment (articles I-IV) Sperm (article I) and hormone analyses (articles II-IV) DNA analysis (articles I, II and IV) Evaluation of sexuality and socio-demographics (article III) Measures of emotional disorders (article IV) Testicular characteristics (article II) Statistical analysis Article I	39 41 41 43 43 43 43 43 43 45 45 45 45 45 45 45 45 46 47 47 48
Aims of the Thesis Material and Methods Patient inclusion Methods Cancer treatment (articles I-IV) Sperm (article I) and hormone analyses (articles II-IV) DNA analysis (articles I, II and IV) Evaluation of sexuality and socio-demographics (article III) Measures of emotional disorders (article IV) Testicular characteristics (article II) Statistical analysis Aims	39 41 41 43 43 43 43 43 43 43 45 45 45 45 45 45 45 45 45 46 47 47 48 48
Aims of the Thesis	39 41 41 43 43 43 43 43 43 43 43 43 43 45 45 46 47 47 48 48 48
Aims of the Thesis	39 41 41 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 45 46 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 47 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48 48
Aims of the Thesis	39 41 41 43 44 48 48 48 48 48 49
Aims of the Thesis	39 41 41 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 45 45 46 47 47 48 48 48 48 48 49 50
Aims of the Thesis	39 41 41 43 43 43 43 43 43 43 43 43 43 43 44 45 45 46 47 47 48 48 48 48 50 50
Aims of the Thesis	39 41 41 43 43 43 43 43 43 43 43 43 43 43 43 43 43 45 45 45 46 47 47 48 48 48 48 48 50 50
Aims of the Thesis	39 41 41 43 43 43 43 43 43 43 43 43 43 43 43 43 43 43 45 45 46 47 47 48 48 48 48 48 50 50 50 50

Article III	53
Aims	53
Patient inclusion and treatment	53
Methods	53
Statistical analysis	54
Article IV	55
Aims	55
Patient inclusion and treatment	55
Methods	56
Statistical analysis	57
Results	59
Impact of therapy and AR polymorphism on sperm concentration	
(article I)	59
Azoospermia	
Sperm concentration	
Factors predicting sperm concentration	61
Correlation between sperm recovery and AR polymorphisms	62
Risk factors for developing hypogonadism (article II)	63
Post-treatment hypogonadism. relation to hypogonadism before treatme	nt63
Hypogonadism in relation to treatment	63
Hypogonadism in relation to age, stage and androgen receptor	
polymorphisms	64
Hypogonadism in relation to testicular characteristics	65
Sexual function and relation to hypogonadism and treatment (arti	cle
III)	66
Sexual function	66
Sexual function in relation to hypogonadism	67
Sexual function in relation to treatment modality	67
EMD in relation to hypogonadism, AR polymorphism and treatme	nt
(article IV)	69
Frequencies of emotional disorders	69
Hypogonadism in relation to EMD	69
AR polymorphisms in relation to EMD	69
EMD in relation to treatment modality	70
Discussion	71
Major findings and clinical implications	71
Spermatogenesis and Leydig cell function	73
Sexual and psychological function	75
Impact of androgen receptor polymorphism	78
Strengths and weaknesses of thesis	80
General conclusions	83
Future Perspectives	
A cknowledgement	20
Deference list	07 07
(iriginal nublications	

Abbreviations

ACT	Adjuvant chemotherapy
AFP	Alpha-fetoprotein
AR	Androgen receptor
β-HCG	beta-humanochoriongonadotrophin
BEP	Bleomycin, etoposide, cisplatin
BEP-if	Bleomycin, etoposide, cisplatin, ifosfamide
BMI	Body mass index
CI	Confidence interval
CIS	Carcinoma in situ
CT	Chemotherapy
DHT	5-alpha-dihydrotestosterone
DNA	Deoxyribonucleic acid
ED	Erectile dysfunction
EMD	Emotional disorders
EP	Etoposide, cisplatin
FISH	Fluorescence in situ hybridization
FSH	Follicle-stimulating hormone
GnRH	Gonadotropin-releasing hormone
Gy	Gray
HADS	Hospital Anxiety and Depression Scale
hCG	Human chorionic gonadotropin
HDCT	> 4 cycles of chemotherapy
HL	Hodgkin's lymphoma
LDH	lactate-dehydrogenase
LH	Luteinizing hormone
Mk+	Elevation of markers (AFP and/or β -HCG)
NSGCT	Non-seminomatous germ cell tumour
OR	Odds ratio
PEI	Cisplatin, etoposide, ifosfamide
RPLND	Retroperitoneal lymph node dissection
RT	Radiotherapy
SCT	3-4 cycles of chemotherapy
SGCT	Seminomatous germ cell tumour
SO	Surgery only
STI	Sexually transmitted infections
SWENOTECA	Swedish-Norwegian testicular cancer project
TC	Testicular Cancer
TDS	Testicular dysgenesis syndrome
TGCC	Testicular germ cell cancer
ТМ	Testicular microlithiasis
WHO	World health organization

List of Original Papers

- I Eberhard J, Ståhl O, Giwercman Y, Cwikiel M, Cavallin-Ståhl E, Lundin KB, Flodgren
 P and Giwercman A. Impact of therapy and androgen receptor polymorphism on sperm concentration in men treated for testicular germ cell cancer: a longitudinal study. *Human Reproduction 2004 Jun; 19(6):1418-25.*
- II Eberhard J, Ståhl O, Cwikiel M, Cavallin-Ståhl E, Giwercman Y, Salmonson EC and Giwercman A. Risk factors for developing hypogonadism in testicular cancer patients. *European Journal of Endocrinology 2008 Apr*; 158(4):561-570.
- **III Eberhard J**, Ståhl O, Cohn-Cedermark G, Cavallin-Ståhl E, Giwercman Y, Rylander L, Eberhard-Gran M, Kvist U, Fugl-Meyer K S, Giwercman A. Sexual function in men treated for testicular cancer. *Accepted for publication (Journal of Sexual Medicine).*
- IV Eberhard J, Ståhl O, Cohn-Cedermark G, Cavallin-Ståhl E, Giwercman Y, Rastkhani H, Rylander L, Eberhard-Gran M, Kvist U, Giwercman A. Emotional disorders in testicular cancer patients in relation to hypogonadism, androgen receptor polymorphisms and treatment modality. *Submitted*.

The papers are reprinted with the permission of the publisher.

Popular scientific summary

The treatment of testicular cancer has become one of the greatest successes in the field of oncology. Since the introduction of cisplatin, in the seventies, the survival rates have dramatically increased and today more than 95 % of patients can expect to be cured.

In view of the fact that the majority of patients are young, between 20 and 40 years of age, more and more attention has been paid to possible sideeffects of the disease and its treatment.

The treatment of testicular cancer may vary in intensity, from surgery, through short chemotherapy or radiotherapy, to intensive cytotoxic treatment. Therefore, the side-effects of the therapy may vary. Contributing to this variation is not only the difference in the treatment given but probably also genetically-determined differences between individuals in their sensitivity to the adverse effects of cancer therapy.

Reproductive function of cancer survivors, including preservation of fertility and normal sexual function, is an important issue. Furthermore, production of sex hormones may influence mental health as well as metabolic and cardiovascular function. In order to improve the counselling of patients regarding their future life quality, research on cancer treatment side-effects in relation to the function of patients' reproductive organs is needed.

The first article focuses on the impact of testicular cancer treatment on sperm concentration in testicular cancer patients. This paper showed that short chemotherapy did not significantly affect sperm production, but men treated by irradiation of lymph nodes in the abdomen, or by chemotherapy for metastatic disease, were at high risk of being sterile six to twelve months after treatment. It seemed, however, that the pre-treatment sperm

production recovered after two to five years, although the size of the study was not sufficient to predict whether the recovery took place in all patients. It means that cryopreservation of sperm should still be offered to all patients before testicular cancer treatment. Interestingly, we found that variation in the androgen receptor gene also influenced the rapidity of post-treatment sperm production recovery. This is, to our knowledge, the first study showing that genetic variation influences the post-treatment recovery of sperm production in cancer patients.

Hypogonadism is a condition related to low levels of the male sex hormone, testosterone. In adult men the symptoms of hypogonadism include low levels of resolution, ambivalence, impaired concentration, fatigue and tiredness as well as osteoporosis, muscle atrophy and loss of sexual desire. The second article has mainly focused on finding predictive factors for hypogonadism, since testicular cancer patients are known to be at risk of developing testosterone deficiency. The article concludes that hypogonadism detected before treatment and certain patterns seen in the remaining testicle by ultrasound are strong predictors of hypogonadism for at least up to five years after treatment. Radiotherapy to abdominal lymph nodes and more than two cycles of chemotherapy were also found to increase the risk of developing hypogonadism, at least transiently, whereas neither age, testicular volume, extension of the disease nor genetic variations in the androgen receptor had any such effect. The findings of this paper will help us in identifying testicular cancer patients at particular risk of developing androgen deficiency.

The third article focuses on sexual dysfunctions in testicular cancer survivors. It concludes that three to five years after treatment, the patients do have significantly increased problems with erectile dysfunction as well as decreased sexual desire. Around 12 % of the patients reported erectile dysfunction compared with 3 % in the general population. Neither different

treatments nor hypogonadism were associated with the risk of having dysfunctions. It means that physicians should focus more on sexual problems in men treated for testicular cancer, although testosterone replacement is not necessarily the treatment of choice. In the fourth article a possible relation between emotional disorders such as depression or anxiety and hypogonadism, treatment intensity and variation in the androgen receptor was evaluated. Neither hypogonadism nor variations of the androgen receptor were associated with the risk of emotional disorders. Patients receiving more than four cycles of chemotherapy owing to refractory or relapsing disease did have a very high (62 %) risk of suffering from anxiety.

The information obtained as a part of this thesis can therefore be applied in the clinical management of testicular cancer survivors by improving the counselling given to these young men and by pointing out some risk factors of serious, life-quality threatening side-effects of cancer treatment.

Background

Testicular cancer

Epidemiology

Testicular cancer (TC) accounts for roughly 1-2 % of all male neoplasms and hence is the commonest cancer disease among males aged between 20 and 40 years (Huyghe *et al*, 2003). For yet unknown reasons, there is a large variation in the incidence between different regions, the highest incidence in the world being reported in Denmark and Norway (Adami *et al*, 1994). The incidence of this cancer has increased four- or fivefold over the last five decades among Caucasian populations for a still unknown reason (Huyghe *et al*, 2003).

Approximately 95 % of the tumours are of germ cell origin (Campbell & Walsh, 1997), i.e. Testicular Germ Cell Cancer (TGCC). Two types of TGCC are distinguished; seminomatous germ cell tumours (SGCT) or nonseminomatous (NSGCT). The distribution between these two histological forms is approximately equal, but the NSGCT patients are slightly younger than those with SGCT with an approximate median age of 27 vs. 34, respectively (Cooper *et al*, 2008).

Aetiology

Aetiological risk factors are still not fully understood even though a history of cryptorchidism as well as atrophic testes and infertility are overrepresented among patients (Raman *et al*, 2005). This has raised the hypothesis of a testicular dysgenesis syndrome (TDS). Since there is rising incidence of testicular cancer (Adami *et al*, 1994; Moller, 1998) and a trend

towards higher incidence of hypospadia (Paulozzi *et al*, 1997) and cryptorchidism (Boisen *et al*, 2004), concomitantly with a decline in semen quality (Moller, 1998; Swan *et al*, 1997), it has also been suggested that all these disturbances in the male reproductive function could be a part of the same syndrome, with a common aetiology and being of foetal origin (Skakkebaek *et al*, 2001).

Diagnosis and staging

TGCC should be suspected in patients who present with a unilateral, intrascrotal, painless mass which can also be detected by ultrasonography. Complementary measuring of tumour markers such as alpha-fetoprotein (AFP), beta-humanochoriongonadotrophin (β -HCG) and lactatedehydrogenase (LDH) is mandatory for disease staging. As the primary treatment, an orchiectomy is performed. Carcinoma-in-situ (CIS) in the contralateral testicle is detected by means of a surgical biopsy. The incidence of contralateral CIS in Denmark and in Germany has been reported to be approximately 5 % (Dieckmann & Loy, 1996; Giwercman *et al*, 1987), but data for other countries are lacking. The orchidectomy is followed by pathological examination of the tumour and a radiological examination for the staging procedure.

For staging, there are different classification systems. In Scandinavia, the Royal Marsden Hospital staging system is most commonly used (Dearnaley *et al*, 2001). At the time of diagnosis, approximately 80 % of SGCT are in stage I (no evidence of metastases) or IIA (metastases to abdominal nodes with a transverse diameter < 2 cm), 20 % presenting with more advanced disease. In SGCT, stage I disease, two risk factors for relapse have been identified, tumour size and invasion of rete testis. For NSGCT about

50 % have detectable metastases at the time of diagnosis (Klepp *et al*, 1990a), but those in stage I have a 30 % risk of relapse if no treatment is given after orchiectomy, which points to a high frequency of subclinical metastases (Klepp *et al*, 1990b). Vascular invasion in the primary tumour is the main risk factor in relapse.

Treatment

According to the recently updated guidelines of the European Consensus Group on testicular cancer, the treatment is dependent on the histology, presence of risk factors and stage of the disease (Krege *et al*, 2008a; Krege *et al*, 2008b). In Sweden and Norway uniform guidelines for treatment have been developed by the Swedish Norwegian Testicular Cancer Project (SWENOTECA) (<u>www.ocsyd.se</u>) and these are principally in accordance with the European guidelines.

In the SWENOTECA protocols, a risk-adapted strategy for stage I disease is applied. NSGCT patients without vascular invasion can choose between one cycle of adjuvant chemotherapy or surveillance.

In case of vascular invasion the recommendation is one cycle of adjuvant chemotherapy. The chemotherapy regimen commonly used is the BEP regimen (bleomycin 30 000 IU days 1, 5 and 15; etoposide 100 mg/m² days 1-5 and cisplatinum 20 mg/m² days 1-5, given every third week). The guidelines for stage I SGCT have recently been changed. Previously, at the time of inclusion of the patients who form part of this thesis, no difference was made between low- and high-risk subjects. All stage I SGCT patients could choose between surveillance and radiotherapy (RT) administered to para-aortic and ipsi-lateral iliacal lymph nodes to a target dose of 25.2 Gy, given in fourteen fractions. The RT is given with lead shield of the remaining testicle. The radiotherapy decreases the recurrence rate from approximately 10% to 2 % (www.ocsyd.se). On the basis of a large randomised trial, the recommendation has now been changed, since one single dose of carboplatin was found to be as effective as adjuvant radiotherapy (Oliver et al, 2005). Today we distinguish between stage I SGCT with no or one risk factor (low risk) and those having two risk factors (high risk), tumour size > 4 cm or invasion to rete testis. Low-risk patients are recommended surveillance and high-risk patients one cycle of CT (carboplatin), RT is still an option, but is considered more toxic than single dose carboplatin and consequently not recommended. In SGCT clinical stage IIA (CSIIA) radiotherapy is recommended, 27.0 Gy given in fifteen fractions against para-aortal and ipsi-lateral iliacal lymph nodes. Disseminated disease is treated with cisplatin-based chemotherapy. The basic strategy of the SWENOTECA protocol for NSGCT is to individualise treatment according to the decline of the tumour markers AFP and β-HCG during the initial treatment. Initial treatment for NSGCT patients is two courses of BEP. Patients with satisfactory response receive one or two additional courses of BEP while patients with unsatisfactory half-time for decline in marker levels receive intensified treatment in two steps with the addition of ifosfamide (BEP-if/PEI) in step I. If there is still unsatisfactory response, the treatment is intensified according to step II, involving highdose chemotherapy with stem cell rescue. Post-chemotherapy surgery, retroperitoneal lymph node dissection (RPLND), is recommended in all patients with abdominal lymph node metastases >= 2 cm at staging and also resection of rest tumours in other locations if possible (Fossa et al, 1989). Standard treatment of SGCT > CSIIA is four courses of EP (=BEP minus B), but treatment intensity is governed by tumour response to the treatment given.

The risk of developing a contralateral cancer is approximately 5 %, which is in accordance with the risk of having cancer in situ (CIS) in a contralateral

testicle (Osterlind *et al*, 1991) and CIS is the precursor for TGCC (Hoei-Hansen *et al*, 2005). Patients with confirmed CIS are recommended radiotherapy, 16 Gy given in eight fractions to the contralateral testicle. With these treatment strategies the prognosis of TGCC patients is excellent with a survival rate exceeding 95 % (Verdecchia *et al*, 2007). Owing to this exceptionally good prognosis the question of long-term toxicity of the treatment and the patient's quality of life has become increasingly important.

Spermatogenesis and male endocrinology

Testicular histology

The testicles contain two functionally different parts: the seminiferous tubules, which contain germ cells and Sertoli cells responsible for sperm production, and the interstitial space, with Leydig cells that are responsible for androgen production.

The hypothalamic pituitary testicular axis

Gonadal function is controlled by the gonadotropin-releasing hormone (GnRH) secreted by the hypothalamus. This hormone stimulates production of the gonadotropins, luteinising hormone (LH) and follicle stimulating hormone (FSH) from the pituitary gland. LH and FSH are secreted in peaks. The stimulation and inhibition of GnRH release seems to be very complex and is not yet fully understood, involving several neurotransmitters and neurohormones, but also with a feedback system from sex steroids. The release is pulsatile and gives a corresponding pulse in LH and FSH release from the anterior pituitary gland (Veldhuis *et al*, 1983). We also know that the pulsatile release of GnRH is crucial for the production of the gonadotropins whereas continuous release actually inhibits the release, a phenomenon which is utilised in pharmacologically-induced castration of e.g. prostate cancer patients (Belchetz *et al*, 1978).

LH targets its receptor on the cell membrane of the Leydig cells and stimulates testosterone production. FSH targets its specific receptor on the Sertoli cell membrane, but the more specific mechanism of initiation of spermatogenesis is still poorly understood.

The secretion of LH is mainly regulated by negative feedback from testosterone, but oestradiol is also involved (Morishima *et al*, 1995).



Figure 1: The hypothalamic pituitary testicular axis. GnRH stimulates secretion of LH and FSH. FSH stimulates spermatogenesis, with a negative feedback from inhibin B. LH stimulates testosterone production. Testosterone inhibits both GnRH and LH release. Testosterone is converted to oestradiol which exerts a negative feedback on the pituitary gland.

Testosterone decreases the frequency and the intensity of the GnRH pulses from the hypothalamus (Matsumoto & Bremner, 1984) and through a direct effect on the pituitary gland (Sheckter *et al*, 1989).

Inhibin B is a peptide produced by Sertoli cells and its production is dependent on the presence of primary spermatocytes and is under the control of FSH (McLachlan *et al*, 1988). Inhibin B is also involved by a negative feedback mechanism in the regulation of FSH secretion. Oestradiol also appears to have a negative feedback on FSH secretion (Hayes *et al*, 2001) (Figure 1).

Androgen production and function

As a precursor steroid, cholesterol, through several enzymatic steps, converts to testosterone in the testes and the adrenal gland (Figure 2). In a post-pubertal male Leydig cells are responsible for 95 % of all testosterone production, the remaining 5 % secreted from the cortex of the adrenal gland. Testosterone, under the control of LH, is not only secreted in a pulsatile fashion (Winters & Troen, 1986), but there is also a significant diurnal variation with peak levels early in the morning, probably sleep induced (Axelsson et al, 2005), and lowest concentration in the evening. In the circulation, most of the testosterone is bound to sex hormone-binding globulin (SHBG) and albumin. The high affinity binding to SHBG accounts for 60-70% of all protein bound testosterone, whereas the low affinity binding to albumin accounts for the remaining 30-40%. Only approximately 2 % of plasma testosterone is in a free form (Dunn et al, 1981). The albumin-bound testosterone is usually bioavailable and therefore the biologically active concentration approximately reflects the summary of albumin-bound and free testosterone. The concentration of the SHBG, which is secreted from the liver, is under the influence of a number of

hormonal factors. The concentration of this protein is decreased by androgen replacement therapy and increased by oestrogen administration. Other factors influencing SHBG levels include thyroid hormone increasing and glucocorticoids, insulin and obesity decreasing the levels of this protein (Gascon et al, 2000; Tchernof & Despres, 2000; Wallace et al, 2003). In healthy, normal men, an increase of SHBG leads primarily to a decrease of free testosterone, but subsequently a reduced negative feedback and increased LH secretion imply a higher level of total testosterone and, thereby, unchanged concentration of the biologically active hormone. In hypogonadal men, the SHBG concentration is usually increased, if the low testosterone levels are not associated with overweight (Selby, 1990). Testosterone is metabolised to 5α -dihydrotestosterone (DHT) and to oestrogens (Figure 2). DHT is the main androgen acting on epididymis, seminal vesicles and prostate, originating from testosterone through 5α reductase. DHT is crucial for normal male sex differentiation in foetal life and also plays a role at puberty in developing the adult male phenotype. The conversion to DHT is organ-dependent and occurs in the target organs. Oestrogens act both synergistically and conversely to androgens and also exert a negative feedback on gonadotropin secretion. In males, oestrogens are mainly produced in extragonadal tissue, in particular, fat tissue. This is one of the explanations for the association between the sex hormone levels and body mass index.

Apart from the role of the androgens in sex differentiation and in initiation of puberty, their main effect is maintenance of spermatogenesis and sexual maturation. These hormones also play a crucial role in the metabolism, muscle and cognitive function and hair growth. Androgens are also important for sufficient mineralisation of the bones, although this effect seems to be mediated through the oestrogens (Kenny & Raisz, 2002; Vanderschueren *et al*, 2000).



Figure 2: Testosterone production and metabolism.

Both testosterone and DHT target the same receptor with high affinity, the androgen receptor (AR). Activation of the AR leads to formation of an active transcriptional regulatory complex that binds with high affinity to androgen response elements in DNA, leading to an up or down regulation of androgen-dependent genes (Lee & Chang, 2003).

The AR is encoded by a gene on the X-chromosome and males, therefore, have only one copy, which they inherit from their mother. The N-terminal domain of the first exon contains a repetitive sequence of CAG triplets encoding a polyglutamine stretch of variable length (Figure 3). In the normal population, the number of CAG repeats varies from approximately ten to thirty (Giwercman *et al*, 1998). There are some ethnic differences, with Africans having shorter and Asians slightly longer repeat lengths than Caucasians (Kittles *et al*, 2001). Among Swedish males, a median number of 22 CAG repeats was reported (Giwercman *et al*, 1998). The repeat length

plays an important role in the transcriptional activity of the AR (Davis-Dao *et al*, 2007; Tut *et al*, 1997; von Eckardstein *et al*, 2001a).



Figure 3: Human AR gene; structural organisation and protein. NTD= Amino-terminal domain. DBD=DNA-binding domain. LBD=Ligand-binding domain

There are reports indicating an association between long CAG repeats and decreased sperm concentrations or infertility (Mifsud *et al*, 2001; Tut *et al*, 1997; von Eckardstein *et al*, 2001a), but these findings have not been confirmed by all studies (Giwercman *et al*, 1998; Rajpert-De Meyts *et al*, 2002).

CAG repeat length has also been found to relate to the risk of obesity and metabolic syndrome although the association is not yet fully clarified (Stanworth *et al*, 2008). It seems that the lower androgen sensitivity in subjects with long CAG is compensated, at least partly, by increased testosterone concentration, probably because of higher LH levels owing to reduced negative feedback through a less sensitive receptor (Stanworth *et al*, 2008). In ageing men, the AR CAG repeat length correlates significantly with testosterone and oestradiol. Weaker transcriptional activity of the AR with longer CAG repeats appears to be totally or nearly totally compensated

for by higher testosterone levels. Consequently, oestrogen levels rise and phenotypic correlations may rather reflect oestrogen than testosterone action (Huhtaniemi *et al*, 2009).

Major expansion of the repeat length up to a length of 40 to 75 repeats causes spinal and bulbar muscular atrophy, known as Kennedy's disease (La Spada *et al*, 1991); the aetiology is not yet fully understood but seems not be related to lower androgen sensitivity but rather to intracellular protein aggregation (McEwan, 2001). Nevertheless, androgen receptors are located throughout the brain (Beyenburg *et al*, 2000; Yaffe *et al*, 2003) and the impact of AR CAG repeat length on neurological and psychiatric disorders has been further investigated. In this context, a possible association between CAG repeat length and affective disorders was postulated, where the severity of depression and anxiety was shown to be negatively correlated with the number of CAG repeats in adolescent patients (Su *et al*, 2007). Depressive symptoms have also been associated with long CAG repeats in ageing males (Harkonen *et al*, 2003).

Another polymorphic sequence in the AR gene is the GGN repeat, encoding a polyglycine stretch. The GGN repeat length varies between 10 and 27, 85 % of the population having 23 or 24 GGN (Lundin *et al*, 2003). A GGN repeat length of 23 is the most common and in vitro (Lundin *et al*, 2007) and in vivo (Lundin *et al*, 2006) studies indicate that this is the length that is associated with optimal AR function.

Spermatogenesis

Spermatogenesis includes a number of steps of cell division and differentiation leading to the development of elongated spermatids from spermatogonia, which are the stem cells of spermatogenesis. Spermatogonia A represent the stem cell pool and the differentiation to spermatogonia B

initiates DNA synthesis resulting in tetraploidic primary spermatocytes. Each primary spermatocyte undergoes a first meiotic division, giving rise to two secondary spermatocytes, which subsequently enter second meiotic division resulting in a total of four haploid spermatids. The next steps include reorganisation of the nucleus and the cytoplasm, development of a flagellum and the head being covered by an acrosomal cap, forming the mature sperm. These last steps are named spermiogenesis. In humans the total duration of spermatogenesis is approximately 70 days. Additionally fourteen days are required for transport of the mature sperm through the epididymis to the ejaculatory ducts. As mentioned above, the hormonal control involves FSH and testosterone acting through their receptors on the Sertoli cells. In the testis, the concentration of testosterone is approximately 100 times higher than in peripheral circulation (Maddocks et al, 1993) and a Sertoli cell selective androgen receptor knockout in mice caused spermatogenic arrest at primary spermatocyte level (De Gendt et al, 2004), showing that AR in Sertoli cells is an absolute requirement for androgen maintenance of complete spermatogenesis, and that spermatocyte/spermatid development/survival critically depends on androgens. Patients with complete FSH receptor mutation were shown, however, to have some sperm production, but the concentration of sperms was very low (Tapanainen et al, 1998).

Exogenous administration of testosterone, in particular when given as injections, leads to inhibition of gonadotropins secretion leading to lack of testosterone production in Leydig cells and thereby azoo- or oligozoospermia (Nieschlag *et al*, 2001).

Germ cells are among the most rapidly dividing cells in the body, which makes them quite sensitive to the effects of CT and RT. Total eradication of spermatogonia B, the first differentiation of the stem cells of spermatogenesis, will lead to permanent azoospermia. This effect has been

seen following more than 4 Gy of irradiation (Rowley *et al*, 1974) and can also be caused by cytotoxic drugs, mainly those with alkylating effect (Howell & Shalet, 2005). If the stem cells are not eradicated, regeneration of spermatogenesis should be expected. In mice, this process usually takes three to six months (Meistrich, 1993), corresponding well to the duration of spermatogenesis. In humans, however, for unknown reasons this process is rather prolonged in some patients, and re-appearance of sperms in the ejaculate has been reported more than five years after completion of cytotoxic treatment (Anserini *et al*, 2002).

Infertility

The definition of infertility is failure to conceive during twelve months of frequent unprotected intercourse (WorldHealthOrganization, 2000). The prevalence of infertility among couples in the western world is approximately

15 %. Around 20 % is reported to be because of a male factor, 38 % to a female factor, 27 % to both male and female factors and 15% is so-called unexplained infertility (WHO Task Force on the Diagnosis and Treatment of Infertility. *et al*, 1987). Thus, in roughly 50 % of all infertility cases some pathology may be found in the male.

The investigation of male infertility includes clinical examination including measurement of testicular volume, semen analysis and hormonal evaluation. Standard semen analysis includes parameters such as ejaculate volume, sperm concentration, motility and morphology assessed according to the WHO criteria (World Health Organization., 1999). Optionally, an evaluation of sperm chromatin can be performed since a high proportion of sperms with chromatin strand breaks is associated with risk of infertility

(Spano *et al*, 2000). Hormonal analysis should include assessment of FSH, LH, testosterone, SHBG, inhibin B and oestrogen.

The causes of male infertility can be divided into hypothalamo-pituitary disease, testicular or post-testicular defects or idiopathic cause. Despite careful assessment, causes of abnormal sperm number, morphology or function cannot be clarified in 40 to 50 % of infertile men (de Kretser, 1997). Testicular disease or primary hypogonadism has been suggested to be responsible for 30 to 40 % of male infertility (de Kretser, 1997).

Testicular cancer, side-effects of disease and its treatment

The high survival rate in TGCC patients during the last few years has implied an increasing attention given to the side-effects of both the disease per se and the treatment given. Thus, the life quality of the survivors has come into focus. As already indicated, the severity of the disease and the treatment modalities differ substantially between patients. In addition, some level of inter-individual variation in the susceptibility to the side-effects of the therapy can be expected (Fry *et al*, 2008). Consequently, frequency, modality and severity of side-effects may vary between patients receiving the same treatment.

When discussing side-effects, we discriminate between acute, transient and/or late side-effects. The most frequent acute side-effects of chemotherapy are nausea, neutropenia and alopecia. Nephropathy, neuropathy and ototoxicity may occur, mainly as a side-effect of cisplatinum and, if these symptoms appear during therapy, there is a risk of chronicity. These side-effects are well investigated and strategies to reduce the risk of developing them are integrated into the routine management of the patients. Another serious late side-effect is the risk of inducing a second

malignancy. This risk has been reported to be two or threefold higher than in the general population (Robinson *et al*, 2007; Travis *et al*, 2005). The risk of developing haematological malignancies is mainly related to etoposide (Nichols *et al*, 1993) and is dependent on the cumulative dose given (Pedersen-Bjergaard *et al*, 1991). Concerning solid tumours, there is an increased risk mainly after RT within the radiation field (Travis *et al*, 2005).

Testicular cancer and infertility

Patients diagnosed with TGCC have an increased risk of being sub- or infertile (Brydoy *et al*, 2005; Moller & Skakkebaek, 1999). The spermatogenesis might be defective already before orchiectomy and deteriorates further after orchiectomy (Petersen *et al*, 1999a; Petersen *et al*, 1999b). The orchiectomy is reported to cause azoospermia in up to 10 % of patients (Petersen *et al*, 1999a).

Furthermore, infertile men have a high risk of developing TGCC. In men presenting with abnormal semen quality, a twenty fold increased risk of developing TGCC was reported (Raman *et al*, 2005). TGCC is also related to cryptorchidism, failure of the testicles to descend to the scrotum during foetal life. Cryptorchidism is associated with poor semen quality and the relative risk of developing TGCC in men with a history of this congenital malformation was reported to be two to ten times higher than in the background population (Dieckmann & Pichlmeier, 2004; Swerdlow *et al*, 1997; Wood & Elder, 2009). Infertility in TGCC patients can also be acquired or accentuated owing to side-effects of TGCC treatment surgery, chemotherapy (CT) and/or radiotherapy (RT).

Overall, approximately 20 % of all treated patients have long-lasting sequelae with infertility or sexual dysfunction (Hartmann *et al*, 1999; Kuczyk *et al*, 2000). Chemotherapy was shown to cause impaired sperm

production and decreased fertility (Bokemeyer et al, 1996; Fossa et al, 1985b; Hendry et al, 1983; Lampe et al, 1997; Pont & Albrecht, 1997; Stephenson *et al*, 1995) and semen quality was found to be significantly more impaired in CT-treated than only orchidectomised patients (Hansen et al, 1990). Following CT, as many as one-third of the men were found to develop at least transient azoo-oligozoospermia (Stephenson et al, 1995). The risk of being permanently azoospermic, however, has been reported to be only between zero and 5 %, but these studies did not include patients receiving more than four cycles of BEP (Gandini et al, 2006; Hansen et al, 1989). The ability of sperm regeneration is dose-dependent. Cumulative doses of cisplatinum exceeding 400 mg/m^2 , equivalent to four courses of BEP regimen, gave long-lasting impairment of gonadal function (Pont & Albrecht, 1997). Nevertheless, after ten years, 38 % of TGCC patients attempting to fertilise, that had received a total dose > 850 mg cisplatin, had achieved paternity without the use of cryopreserved sperm. These figures increased to 48 % after fifteen years (Brydoy et al, 2005). Adjuvant RT to abdominal lymph nodes might also impair sperm production owing to scattered radiation to the remaining testicle. But if scattered doses can be limited <0.2 Gy permanent radiation-induced effects on the remaining testicle are very unlikely (SedImayer et al, 1999). In the adjuvant setting and RT given by modern technique, the risk of permanent azoospermia is very small, current reports mainly pointing towards a transient effect (Centola et al, 1994; Sedlmayer et al, 1999). Radiation doses exceeding 4 Gy have been reported to induce permanent azoospermia (Rowley et al, 1974). Direct radiotherapy to the remaining testicle with doses of 14-20 Gy given for eradication of CIS implies permanent azoospermia (Giwercman et al, 1991).

Retroperitoneal lymph node dissection (RPLND) (Fossa *et al*, 1985a) is a well-known cause of retrograde ejaculation. With a unilateral nerve-sparing

procedure, introduced during the 1980s, the risk of retrograde ejaculation varies from a few percent to almost 30 % (Donohue, 2003; Jacobsen *et al*, 1999; Krege *et al*, 2008a; Krege *et al*, 2008b), while with more extensive bilateral surgery the risk is very high.

Since it is not completely predictable which men will develop long-standing or permanent azoospermia, sperm cryopreservation is routinely recommended to TGCC patients. In a large study on TGCC survivors, approximately 50 % were interested in pre-treatment semen cryopreservation (Magelssen *et al*, 2005). A considerable number of these achieved fatherhood without the use of frozen semen, but the psychological impact of pre-treatment cryopreservation was undeniable. For 7 % of the survivors, however, assisted reproductive techniques with cryopreserved sperm offered the only chance of post-treatment paternity (Magelssen *et al*, 2005).

Several studies have addressed the issue of recovery of spermatogenesis following TGCC treatment and whether it could be predicted or not (Aass *et al*, 1991; Fossa *et al*, 1990; Lampe *et al*, 1997). Pre-treatment decreased gonadal function (low sperm count and increased FSH) is a risk factor of persistent testicular dysfunction (Fossa *et al*, 1990; Lampe *et al*, 1997). Other pre-treatment risk factors include defective sperm chromatin structure as well as age (Aass *et al*, 1991; Fossa *et al*, 1997). Since androgens play an important role in spermatogenesis and the

physiological effect of testosterone in humans is modified by the length of CAG and GGN repeats of the AR gene, it could be speculated that these polymorphisms might also play a role in the regeneration of sperm production following cancer therapy.

Although several studies have addressed the issue of semen quality and fertility in TC survivors, the current literature has several limitations. Many

studies do not discriminate between the effects of different treatment modes including CT, RT, orchiectomy and RPLND (Fossa *et al*, 1985b; Hansen *et al*, 1990; Hendry *et al*, 1983; Joos *et al*, 1997; Lampe *et al*, 1997; Pont & Albrecht, 1997; Stephenson *et al*, 1995) and do not take the intensity of treatment into account (Bokemeyer *et al*, 1996; Hansen *et al*, 1990; Stephenson *et al*, 1995). Also, knowledge about the time course is still scarce. Furthermore, the issue of genetically-determined inter-individual variation in post-treatment recovery of spermatogenesis has not yet been investigated.

To be able to give adequate information to TGCC patients about future fertility, there is a need for longitudinal studies taking into consideration treatment modality, length of follow-up period and also identification of possible genetic markers of restoration of sperm production.

Testicular cancer and hypogonadism

The diagnosis of hypogonadism is based on a combination of biochemical and clinical features.

In men with symptoms of androgen deficiency, the combination of high gonadotropin levels and low testosterone indicates testicular origin of hypogonadism or primary hypogonadism, which is the predominant type in TGCC patients.

The clinical symptoms of hypogonadism include: loss of libido, erectile dysfunction, impairment of memory, depression, lethargy, osteoporosis, loss of muscle mass and strength and some regression of secondary sexual characteristics, commonest in post-pubertal men (Carnegie, 2004). Furthermore, it has been shown that hypogonadism is a risk factor for development of metabolic syndrome and cardiovascular disease (Kupelian *et al*, 2006) and TGCC patients have an increased risk of both

cardiovascular risk profile and manifest disease (Huddart *et al*, 2003; Nuver *et al*, 2005b; Vaughn *et al*, 2008; Wethal *et al*, 2007). The pathophysiology behind this association is still not fully understood and different mechanisms have been postulated. It has been speculated that this effect is related to a direct negative effect of chemotherapy on blood vessels (Nuver *et al*, 2005a) and may also be a consequence of impaired renal function (Bosl *et al*, 1986). A link between hypogonadism and cardiovascular disease should also, however, be considered.

TGCC patients are at risk of developing hypogonadism (Nijman et al, 1987; Willemse *et al*, 1983). The disease per se seems to be associated with Leydig cell insufficiency, since orchidectomised men with TGCC have lower levels of testosterone compared with orchidectomised non-TGCC patients (Willemse et al, 1983). Orchidectomy is followed by a decrease in testosterone levels (Fossa et al, 1984) but a time-related compensatory improvement may occur, which is why hesitation in initiating replacement therapy shortly after surgery is warranted (Petersen et al, 1999a). Chemotherapy and radiotherapy might affect Leydig-cell function negatively. A total of 60 % of TGCC patients receiving cisplatinum-based CT owing to metastatic disease had elevated LH levels compared with 11 % of those not given CT several years after treatment (Hansen et al, 1990). The negative effect of cisplatin-based chemotherapy with persistent LH elevation and low testosterone is further confirmed in other investigations (Fossa et al, 1995; Hansen & Hansen, 1993; Strumberg et al, 2002). There are, however, some studies where no association between cisplatin-based CT and Leydig cell dysfunction could be found (Fossa *et al*, 1985b; Petersen *et al*, 1994). These dissimilarities might be because of differences in chemotherapy used, doses and length of follow-up period. The impact of adjuvant sub-diaphragmal RT is less studied but a possible transient increase of LH, following this treatment, has been reported (Joos et

al, 1997). Leydig cells are, however, generally considered to be less sensitive to irradiation than the germinal epithelium.

RT to the contralateral testis is given when cancer in situ is diagnosed. Endocrine function seems to be impaired already before treatment with further impairment after testicular irradiation with 14-20 Gy (2 Gy x 7-10) but only with minor dose dependency in the range of 14 to 20 Gy (Petersen *et al*, 2003; Petersen *et al*, 2002). An unresolved issue is whether the longterm age-dependent decrease in testosterone levels is accentuated by this Leydig cell damage.

Despite the fact that many studies have focused on testosterone and LH levels in TGCC-treated men, little is known about risk factors of developing hypogonadism. In a ten-year follow-up study, men treated for TGCC had a three- to fourfold risk of developing hypogonadism and the risk increased with age and treatment intensity (Nord *et al*, 2003). The predictive value of treatment intensity was further confirmed where both radiotherapy and chemotherapy resulted in additional Leydig cell impairment compared with surveillance only (Huddart *et al*, 2005).

Apart from the treatment-related factors, however, other characteristics might also be associated with the risk of testicular dysfunction and hypogonadism.

Testicular microlithiasis (TM), detected by ultrasonography, is overrepresented among TGCC patients, found in men during infertility investigation (Costabile, 2007). Thus, one could speculate whether men presenting with TM also have an increased risk of developing posttreatment hypogonadism. Another factor of potential interest is the genetically-determined sensitivity to androgens, related to the polymorphisms in the AR gene (Lundin *et al*, 2006; Tut *et al*, 1997). Bearing in mind that the symptoms of hypogonadism are rather uncharacteristic and therefore may be overlooked during the follow-up, it would be of help to define men at high risk. Characteristics such as age, pretreatment hormone levels, treatment modality, stage of disease, presence of TM, contralateral testicular volume and genetic markers should, therefore, be further investigated to identify risk factors for androgen deficiency in TGCC patients.

Testicular cancer and sexual dysfunction

The commonest male sexual dysfunctions are erectile dysfunction (ED), ejaculatory dysfunction and decreased sexual desire or interest (Halvorsen & Metz, 1992a). The prevalence in the general population has been reported as 4 to 9 % for ED, 4 to 10 % for absent or delayed ejaculation and 36 to 38 % for premature ejaculation (Spector & Carey, 1990). There are studies suggesting an increase in the frequency of absent or delayed ejaculation as well as erectile dysfunction and a decrease in premature ejaculation. Desire disorders have increased in prevalence among patients referred for sexological treatment (Spector & Carey, 1990). The aetiology of different categories of sexual dysfunction differs substantially and includes neurological, vascular, endocrinological, psychological, traumatic or pharmaceutical causes. The anamnesis could give a hint about the underlying cause; loss of nocturnal erection indicates a neurological or vascular cause, sudden onset a traumatic or psychological reason and an unsustained erection might indicate a vascular or psychological cause (Halvorsen & Metz, 1992b).

Metabolic syndrome is a risk factor for cardiovascular disease, including reduced penile blood flow with subsequent ED (Fung *et al*, 2004), and since TGCC patients have an increased risk of developing metabolic syndrome (Wethal *et al*, 2007) it could be hypothesised that these patients also have an

increased risk of ED. There are even reports indicating that ED could predict later coronary heart disease (Min *et al*, 2006; Thompson *et al*, 2005). TGCC patients are at risk of sexual dysfunctions, the most established being retrograde ejaculation owing to RPLND surgery (Fossa *et al*, 1985a). Many patients consider this side-effect as a more important problem in relation to infertility then its impact on sexual life. Today, there are different options to assist these patients in achieving fatherhood, including use of α adrenerg receptor stimulators (Ochsenkuhn *et al*, 1999), penile vibration, trans-rectal electroejaculation (Ohl *et al*, 1991) or use of testicular or epididymal sperm extraction followed by in vitro fertilisation or intracytoplasmic sperm injection (Rosenlund *et al*, 1998). Another option is using cryopreserved sperms.

Among TGCC patients, however, several other sexual dysfunctions have been reported and, apart from ejaculatory dysfunction, erectile dysfunction (Jonker-Pool *et al*, 2001) and absent or reduced orgasm are overrepresented (Nazareth *et al*, 2001), which is also true for decreased sexual enjoyment and desire (Joly *et al*, 2002).

Since the aetiology of sexual dysfunctions differs substantially, a number of risk factors have been evaluated. High age at treatment (Aass *et al*, 1993; Caffo & Amichetti, 1999) as well as treatment modality, chemo- or radiotherapy (Jonker-Pool *et al*, 1997; Tinkler *et al*, 1992), has been reported as having a negative impact on sexuality. Recovery over time has been seen, more patients reporting dissatisfaction with sexual life six months after therapy than before treatment, but with some recovery after three years (Aass *et al*, 1993).

Another aspect of importance in TGCC patients with sexual dysfunctions is the issue of psychological stress caused by the threat from a malignant disease (Jonker-Pool *et al*, 2001). An additional factor to be taken into consideration is hypogonadism, since both TGCC and sexual dysfunction
have associations with androgen deficiency. There is, however, a lack of studies focusing on this association. Although Wiechno and colleagues (Wiechno et al, 2007) found significant association between increased LH levels and symptoms of sexual dysfunction as assessed by use of the Sexual Functioning Questionnaire (SFQ), low testosterone levels were not found to be a risk factor for abnormal SFQ or erectile dysfunction assessed by the International Index of Erectile Function (IIEF) (Wiechno et al, 2007). In most reports, as a methodological tool global or domain-specific questionnaires were used to assess sexual function, giving composite scores. Item by item comparison with a control group has not been done. Longitudinal reports are also scarce. Owing to these methodological shortcomings, full knowledge about risk factors and frequency of sexual dysfunctions in TGCC patients is still lacking. Consequently, since sexual function is an issue of great importance to many patients, further investigations are warranted in order to improve knowledge about the cause of this problem. Such information is important for prevention, early detection and also treatment of sexual dysfunction in TGCC survivors. Thus, androgen deficiency can easily be treated by replacement therapy, whereas sexual problems not related to lack of testosterone require other management strategies.

Testicular cancer and emotional disorders

In oncological care the main objective concerning psychiatric co-morbidity is to identify the major general diagnoses.

According to the standardised psychiatric diagnose system, diagnostic and statistical manual of mental disorders (DSM IV) (American Psychiatric Association, 1995) there are several emotional disorders (EMD), however, in current thesis only anxiety disorders and depression are discussed as EMD.

The lifetime risk of having a mood disorder or depression is approximately 20 % (Kessler *et al*, 1994) and the risk within a twelve-month period of having an anxiety disorder is 18 % whereas it is approximately 9 to 10 % for depression (Kessler *et al*, 2005; Kroenke *et al*, 2007). Also, in developed countries world-wide mental illness is reported to count for about 15 % of total disability (Murray & Lopez, 1997).

A recent report observes that the non-psychiatrist physician's accuracy in recognising depression is low and that it is necessary to develop methods to improve the physician's ability to recognise depression (Cepoiu *et al*, 2008).

The diagnostic criteria for major depression according to DSM IV are presentation of at least five of the following symptoms most of the day, nearly every day, for at least two repeated weeks:

- depressed mood;
- loss of interest/pleasure;
- change in sleep;
- change in appetite or weight;
- change in psychomotor activity;
- trouble concentrating;
- loss of energy;
- thoughts of worthlessness or guilt;
- thoughts about death or suicide.

Sometimes patients have fewer than five of these symptoms or the symptoms are lighter, which could then be characterised as minor depression or dysthymic disorder.

Anxiety disorders can be divided into several different diagnoses, such as panic disorder, different phobias including social phobia, generalised

anxiety disorder, obsessive-compulsive disorder and post-traumatic stress disorder, according to DSM IV (American Psychiatric Association, 1995). Accordingly, different anxiety diagnoses have their specific diagnostic criteria, but some of them are common to all anxiety disorders such as excessive, irrational fear and dread. Other common symptoms for many of the disorders are unfounded worry, irritability, sleeping disorders, feeling nervous, psychosomatic symptoms such as palpitations, difficult in breathing, dizziness and chest pain.

Several reports indicate that these psychiatric disorders are underdiagnosed (Kroenke et al, 2007; Wittchen et al, 2001; Wittchen et al, 2002) as well as being very common in society (Kessler et al, 2005; Kessler et al, 1994; Kroenke et al, 2007). To reveal a correct diagnosis the best way is still structured face-to-face questioning based on the DSM IV criteria, but in the hope of facilitating the diagnostic procedure and also to screen for the disorders, several questionnaires have been developed. A number of these tools are well-validated and applicable in different clinical situations. One of the most extensively validated and commonly used is the Hospital Anxiety Depression Scale (HADS). The HADS questionnaire consists of fourteen questions, seven concerning depression (HADS-D) and seven concerning anxiety (HADS-A). Each question is scored from 0 to 3 and the score on each subscale (HADS-A and HADS-D) ranges from 0 to 21. HADS-A≥8 indicates anxiety, HADS-D≥8 indicates depression (Zigmond & Snaith, 1983). A review based on 747 papers showed sensitivity and a specificity of approximately 0.80 for both anxiety and the depression-related part of HADS (Bjelland et al, 2002). Furthermore, a comparative validation between HADS and Structured Clinical Interview for DSM IV (SCID) showed that by use of the recommended cut-off points for HADS a sensitivity of 85 % and a specificity of 76 % for diagnosing EMD were achievable (Lowe et al, 2004).

Even though TGCC patients report good general satisfaction with life (Fleer et al, 2004; Joly et al, 2002), these men have an increased risk of anxiety (Dahl et al, 2005b; Fossa et al, 2003) and also, if presenting with chronic fatigue, there is an association between both depression and anxiety following successful cancer treatment (Dahl et al, 2005a). Little is known, however, about possible predictive causes. Treatment modality is one possible cause but no association has been found between the different treatment modalities and risk of reduced quality of life (Mykletun et al, 2005). Hypogonadism and depression have several symptoms in common such as low level of resolution, ambivalence, impaired concentration, fatigue and low energy serving an obvious problem in making the right diagnose. A possible association between depression and biochemical hypogonadism has been reported, where high LH levels were associated with increased risk of being depressed (Wiechno et al, 2007). In this context, another interesting observation was a study on ageing males that pointed to a higher risk of depression in men with long CAG repeats (Harkonen et al, 2003), while in another investigation on adolescent men, short CAG repeats pointed to not a higher incidence, but more severe depressive symptoms (Su et al, 2007). Additionally, greater CAG repeat length was associated with lower scores on cognitive tests, probably also implying an association with EMD (Yaffe et al, 2003).

In general, although several studies have addressed the issue of long-term sequelae of TGCC and its treatment, there is still a significant lack of information regarding the predictive factors for such complications and also the strategies for their prevention. These obstacles represent a serious hindrance in optimal management of TGCC survivors.

Aims of the Thesis

The overall aim of this thesis was to increase the current level of knowledge regarding impairment of reproductive function and risks of emotional disorders related to TGCC and its treatment and thereby improve the management and counselling of this group of men.

The specific aims of the individual studies were, in TGCC patients, to:

- assess the effect of CT and RT on semen quality with special attention to the dose response effect and the time course of recovery;
- investigate the impact of different AR polymorphisms on pretreatment sperm characteristics and as a predictor of sperm regeneration after treatment;
- identify risk factors for developing androgen deficiency following cancer treatment;
- estimate the prevalence and characterise the type of sexual dysfunction, three to five years after therapy;
- evaluate the impact of hypogonadism, genetically determined androgen sensitivity and treatment intensity in relation to the risk of post-treatment sexual dysfunction and emotional disorders.

Material and Methods

Patient inclusion

All TGCC patients referred to the Department of Oncology, Lund University Hospital, Lund between the ages of 18 and 50 and diagnosed within a period of five years prior, were asked to participate in a study of fertility.

The study was initiated in March 2001. In November 2003 additional inclusion started at the Department of Oncology, Radiumhemmet and Södersjukhuset, Karolinska University Hospital, Stockholm.

The inclusion was discontinued in June 2006, and as the patients are followed for longitudinal investigation for five years the study will go on until 2011.

Of 461 eligible patients, 334 were included (72 %). Seventy-five patients declined participation (16 %) and 52 patients were exclude due to mental co-morbidity, linguistic difficulties, bilateral testicular cancer or physically disabled (Figure 4).

Article I and II only included patients from Lund, article III and IV from both Lund and Stockholm.

All patients participated with a written informed consent according to protocols approved by the ethical review boards of Lund University.



Figure 4: Flow-chart describing the patient material. Article I and II, being longitudinal and III and IV cross-sectional.

Additionally two patients eligible for study III (n=336).

Methods

*

From included patients were obtained:

- clinical information regarding histological diagnose, stage of disease and cancer treatment;
- semen samples;
- blood samples for hormone analyses;
- blood sample for DNA analysis;
- ultrasound of contra-lateral testicle;
- questionnaire concerning sexual function, socio-demographics, quality of life and emotional disorders.

Cancer treatment (articles I-IV)

All patients were treated according to the SWENOTECA protocols (Albers *et al*, 2005; Klepp *et al*, 1997), the details described on pages 15-17. For staging, the Royal Marsden Hospital (RMH) staging system was used (Dearnaley *et al*, 2001; Horwich *et al*, 1989).

Sperm (article I) and hormone analyses (articles II-IV)

Six time-points for delivery of both ejaculates and blood samples for hormones were defined (Figure 5):

- after orchiectomy but before further treatment;

- 6, 12, 24, 36 and 60 months after completion of treatment.

The ejaculates were analysed according to the WHO (1999) manual and sperm motility, morphology, and concentration as well as ejaculate volume were determined (World Health Organization., 1999). For those recruited in Lund, all sperm analyses were performed at the Reproductive Medicine Centre (RMC) (former Fertility Centre), Malmö University Hospital, Malmö. Patients from Stockholm were not included in this part of the study (article I). A few cryopreserved samples, collected prior to inclusion, were analysed in the fertility laboratory, Lund University Hospital, Lund.

Hormone analyses included luteinizing hormone (LH) and testosterone. Blood sampling was performed between 8 am and 3 pm. All analyses were performed in one of three laboratories, the Departments of Clinical Chemistry in Malmö, Lund and Stockholm. During the study period, the methods for LH and testosterone analyses were changed in Lund and Malmö and conversion factors were obtained, both for the intra- and interlaboratory variation. In a pooled dataset, no difference in hormone levels measured by the different laboratories was found.

The reference levels for testosterone and LH were identical between Malmö/Lund and Stockholm. Patients were categorized as being hypogonadal if serum testosterone was below 10 nmol/L and/or serum LH was 10 IU/L or more (Nieschlag *et al*, 2004). Since the blood samples were taken between 9 am and 3 pm and the levels of testosterone, but not LH, decrease during the day, we also used LH>10 IU/L as the only indicator of hypogonadism.

Depending on time from diagnosis and inclusion in the study, a patient could contribute with one to six samples during the study period (Figure 5).



Figure 5: *Study Design. The circle represents semen/blood sampling, the black square time of inclusion.*

DNA analysis (articles I, II and IV)

Androgen receptor CAG and GGN repeat lengths were analysed in DNA extracted from peripheral leukocytes and amplified by polymerase chain reaction. Subsequently, direct sequencing was done using Beckman Coulter CEQ 2000XL (Beckman Coulter, Bromma, Sweden) sequencing gear (Lundin *et al*, 2003).

Evaluation of sexuality and socio-demographics (article III)

In 1996 a nation wide Swedish study on sexual life in Swedes age 18 to 74 years was performed by the Swedish National Institute of Public Health (Fugl-Meyer *et al*, 2000). The questionnaire used in this national study was the source of the questionnaire in the preset study.

Questions relevant for our study population were selected by a multidisciplinary group, involving one psychiatrist/sexologist, one andrologist and two oncologists. From the national questionnaire, nineteen questions were chosen reflecting:

- socio-demographic status including information regarding having/not having a partner, paternity status, sexually transmitted infections (STI), alcohol, smoking and snuffing habits, weight and height;
- sexual functions/dysfunctions with propensity on sexual desire, sexual interest, erectile dysfunction (ED), personal problem due to ED, premature and delayed ejaculation, time since last intercourse, sexual satisfaction and need for sexual advice;
- general satisfaction.

At two time-points, at time of the inclusion and after 60 months, all men filled out the questionnaire. As reference population, 916 age-matched men, who had participated in the national study, were used.

Measures of emotional disorders (article IV)

At the time of inclusion and after 60 months, the patients filled in the hospital anxiety depression scale (HADS), which is a well established and validated self-rating scale developed to screen for depression and anxiety (Zigmond & Snaith, 1983). HADS contains fourteen items; seven depression items (HADS-D) and seven anxiety items (HADS-A). Each item is scored from 0-3 and the score on each subscale (HADS-D and HADS-A) ranges from 0 to 21. A HADS-D score ≥8 was used as the cut-off score for depression and HADS-A \geq 8 was used for anxiety. These cut-off levels correspond to those used in prior studies (Bjelland *et al*, 2002).

Testicular characteristics (article II)

The remaining testicle was evaluated with ultrasound concerning volume and presence of microlithiasis (TM) between six and twelve months after inclusion of the patient. For determination of testicular volume, the formula for volume of ellipsoid: $1/3 \ge \pi \ge 1000$ km s wide $\ge 1/2$ was used (Lenz *et al*, 1993). TM was defined as at least five uniform, non-shadowing echogenic foci of 1 to 3 mm, scattered throughout the testicular parenchyma (Lenz *et al*, 1987; von Eckardstein *et al*, 2001b). All examinations were made by the same radiologist. The consistency was investigated by palpation and assessed as being normal or soft.

Statistical analysis

All statistical analysis was performed using the SPSS 11.0 software (SPSS Inc., Chicago, USA). For all statistical tests p<0,05 was considered statistically significant.

Details of material and methods are presented in the original articles of the thesis. For each article, a summary is presented below.

Article I

Aims

To evaluate the impact of different TGCC treatment modalities on semen quality, focusing on dose response effect and time course of recovery.
To evaluate the impact of different AR polymorphisms on pre-treatment sperm concentration and as a predictor of sperm regeneration after treatment.

Patient inclusion and treatment

Until April 1st 2003, 112 of 144 eligible patients (78 %) were recruited (Figure 4).

Details on treatment are given in pages 15-17 and table 1. Eleven patients developed retrograde ejaculation due to retroperitoneal lymph node dissection (RPLND). For details on number of semen samples delivered postorchidectomy and at follow-up, see figure 4. Eight of the 112 included patients had not yet delivered a semen sample at the time of study evaluation.

Methods

Sperm and DNA analyses are described on page 43.

When the analyses were initiated, blood samples for DNA analysis were available only from the first 81 men. The number providing semen samples were 56, 23, 42 and 31 at pre-treatment, six months, one to two years and three to five years, respectively. This subgroup of men for whom the CAG and GGN repeat lengths could be determined, did not differ from the remaining 31 with respect to age, disease, histological type, stage, treatment or sperm concentration post-orchidectomy.

Statistical analysis

For each treatment modality, a longitudinal analysis of data was performed. In order to obtain sufficient numbers, samples collected at 24, 36 and 60 months were pooled. If a patient delivered more than one sample during this time interval, the one with the highest sperm concentration was included in the analysis.

Additionally, comparison of semen parameters between groups, a crosssectional analysis was performed.

For longitudinal comparisons of more than two time points Friedman's test was used. For intra-individual comparison of values at two time points only, Wilcoxon test for paired data was applied. In the cross-sectional analyses, Kruskal-Wallis test and Mann-Whithey test for unpaired data were used. Spearman's rho was calculated in order to find the correlation between the CAG or GGN repeat length and the sperm concentration at any of the following time points:

- after orchidectomy but before further treatment;
- six months, one to two years and three to five years.

DNA data were caculated for the whole group as well as separately for the therapy groups ACT, SCT and RT.

Subsequently, in order to calculate the predictive value on the pre-treatment sperm concentration, multivariate linear regression analysis was used with the type of tumour, as discrete variable, and age and CAG repeat length as continuous variables,. Similar type of analysis was done for sperm

concentration at six months, one to two years and three to five years, with type of therapy as discrete variable and CAG repeat lengths and age as well as sperm concentration pre-treatment as continuous independent variables. Sperm concentrations were log transformed (after adding 0.1 to all sperm concentrations in order to be able to transform 0 values) prior to the analysis. All statistical tests were two-sided.

Article II

Aims

-To identify patient, disease and treatment related risk factors associated to and predicting post-treatment hypogonadism.

Patient inclusion and treatment

Until December 31st, 2005, 149 of 200 eligible patients (74 %) were included and 143 were evaluable (six patients had not delivered any blood samples) (Figure 4).

Details on treatment are given in pages 15-17 and table 1. For details on number of samples for hormone analyses delivered post-orchidectomy and at follow-up, see figure 4.

Methods

Hormone and DNA analyses as well as ultrasound of the testicle are described on pages 41-43 and 45.

Blood samples for DNA analysis were, as in study I, available from 81 men. These 81 subjects did not differ from the remaining 62 regarding age, disease, histological type, stage, treatment or sperm concentration postorchidectomy. Of the 143 patients, 107 underwent an ultrasound of the remaining testicle.

	Article I						Article II					
Treatment modality	SO	ACT	SCT	HDCT	RT	All	so	ACT	SCT	HDCT	RT	All
No of patients	7	32	37	5	31	112	13	43	42	9	36	143
Age (median)	24	28	28	28	35	29	29	29	27	29	36	30
SGCT	0	0	7	2	31	40	5	2	8	3	36	54
NSGCT	7	32	30	3	0	72	8	41	34	6	0	89
Stage I	7	32	7	0	31	77	13	43	9	2	36	103
Stage II	0	0	20	1	0	21	0	0	21	2	0	23
Stage III	0	0	4	1	0	5	0	0	4	2	0	6
Stage IV	0	0	6	3	0	9	0	0	8	3	0	11

Table 1: Treatment of patient in article I and II.

Royal Marsden Hospital (RMH) staging system has been used (Horwich etal, 1989)SONo further therapy after orchidectomy;ACT1-2 cycles of adjuvant chemotherapy;SCT3-4 cycles of chemotherapy;HDCTMore intense treatment, >4 cycles of chemotherapy;

RT Adjuvant radiotherapy;

Statistical analysis

Patients with an increased level of human choriogonadotropin (HCG) were excluded from the analysis of post-orchidectomy hypogonadism. Seven men being on androgen replacement therapy were considered hypogonadal from the time the replacement was given. Two of them were in the ACT group, two in the SCT group, two in the RT group and one was in the HDCT group.

Primarily, using binary logistic regression analysis, OR for hypogonadism was calculated for TGCC patients using the ACT group as a reference group. A separate comparison was done for each treatment modality (SCT/RT) and post-treatment time point. Furthermore, the risk for the whole group of TGCC men for developing hypogonadism was tested at different post-treatment time points, with respect to the following potential predictive factors:

- biochemical hypogonadism before treatment (yes/no);
- androgen receptor CAG (<21, 21-22, >22) and GGN (<23, 23, >23) repeat number;
- stage of disease (I-IV);
- testicular consistency (normal/soft);
- ultrasound (+/- microlithiasis);
- volume of the contralateral testis (<15ml vs.≥15ml);
- age (<30 vs.≥30 years).

Patients presenting with TM, were complementary analysed after excluding those being hypogonadal postorchidectomy, but before further treatment. The method of hormone analysis was used as potential confounder. All analyses were performed separately for each time point.

Article III

Aims

-To study the prevalence and type of sexual dysfunctions 3 to 5 years after treatment for TGCC and compare these figures to those in an age-matched group of men from the general population.

-To relate our findings to type of oncological treatment and presence of hypogonadism among TGCC patients.

Patient inclusion and treatment

Until 30th of June 2006, 334 patients were included in the fertility study. Additional two patients, not eligible for the fertility study due to bilateral TGCC were considered eligible in study III (n=336). Of these patients 129 had past the three year control and were thus included in study III (Figure 4). Details on treatment are given in page 15-17 and table 2. Eight patients had retrograde ejaculation after RPLND.

Methods

Evaluation of sexuality and socio-demographic as well as hormone analyses are described on pages 41-44. For the sexual functions/ dysfunctions a sixgraded answering alternative scale was used:

- 1. never;
- 2. hardly ever;
- 3. rather rarely;
- 4. rather often;
- 5. nearly all the time;
- 6. all the time.

A patient who stated that the dysfunction occurred rather often/nearly all the time/all the time was considered to suffer from manifest dysfunction per se (Fugl-Meyer & Fugl-Meyer, 2002). If the dysfunction led to personal erectile distress the same scale was used.

Times since last intercourse was dichotomised in two different ways, primarily with an approximate 50 % distribution (less than 5 days vs. 5 days or more), secondly with an approximate 10 % distribution (3 months or less vs. more than 3 months)

Sexual satisfaction was assessed by the question "How satisfying is your sexual life?" derived from the well-validated generic instrument LiSat-11 checklist (Fugl-Meyer *et al*, 2002). Six answering alternatives ranging from very dissatisfied to very satisfied were offered. The scale is test-retest reliable and it is valid to dichotomize the scale into "satisfied" (very satisfying or satisfying) and "not satisfied" (rather satisfying/rather dissatisfying/dissatisfying/very dissatisfying).

Statistical analysis

To evaluate the likelihood probability for co-occurrence of TGCC and sexual dysfunctions, OR with 95 % CI were calculated, using logistic regression. All analyses were adjusted for age. In addition, potential confounders such as occupation, paternity status, failing to become biological father, smoking, snuffing, sexually transmitted infections and BMI, were included in the models, one at a time. These items were kept in the model if they changed the age-adjusted effect estimate with more than 15%. Patients reporting absence of antegrade ejaculation was excluded from the analyses concerning premature and delayed ejaculation need for sexual advice as well as for the question regarding consulting an expert.

The impact of different therapeutic modalities in relation to indices of sexual dysfunction was assessed by comparing them to each other. We used SCT as reference since it was the largest group and as we, a priori, expected the SCT treated men to be the most seriously affected. For all statistically significant associations between disease/treatment and sexual function outcomes, biochemical hypogonadism as predictor of sexual dysfunction was then tested, using binary logistic regression. For the variables analysed by binary logistic regression, the outcomes were dichotomised. Patients on testosterone replacement (n=9) were excluded from this analysis.

Article IV

Aims

-To evaluate whether biochemical signs of hypogonadism and/or AR polymorphisms are predictors of emotional disorders (EMD) in TGCC survivors.

-To assess the association between treatment intensity and EMD.

Patient inclusion and treatment

Until 30th of June 2006, 334 patients were included in the fertility study. One-hundred and sixty-five patients who went through their three-year check-ups were included in the study (Figure 4). Details on treatment are given in pages 15-17 and table 2.

	Article III						Article IV					
Treatment modality	SO	ACT	SCT	HDCT	RT	All	SO	ACT	SCT	HDCT	RT	All
No of patients	11	33	47	2	36	129	13	41	54	8	49	165
Age (median)	32	33	33	40	37	35	32	35	35	34	40	36
SGCT	3	0	15	1	36	55	4	0	16	3	49	72
NSGCT	8	33	32	1	0	74	9	41	<i>38</i>	5	0	<i>93</i>
Stage I	11	33	10	1	36	91	13	41	12	1	49	116
Stage II	0	0	25	0	0	25	0	0	28	2	0	30
Stage III	0	0	4	0	0	4	0	0	4	1	0	5
Stage IV	0	0	8	1	0	9	0	0	10	4	0	14

Table 2: Treatment of patients in article III and IV.

Royal Marsden Hospital (RMH) staging system has been used (Horwich et al, 1989)

SO	No further therapy after orchidectomy;
ACT	1-2 cycles of adjuvant chemotherapy;
SCT	3-4 cycles of chemotherapy;
HDCT	More intense treatment, >4 cycles of chemotherapy;
RT	Adjuvant radiotherapy;

Methods

Evaluation of emotional disorders as well as hormone and DNA analyses are described on pages 43-45. Blood samples for DNA analysis were available from 140 men for CAG repeat length and from 135 on GGN repeats. These subjects did not differ from the remaining 25/30 regarding age, disease, stage, histological type, biological hypogonadism or prevalence of EMD.

Statistical analysis

Using binary logistic regression, OR with 95 % CI were calculated to evaluate the association between biochemical signs of hypogonadism, length of CAG repeat, and length of GGN repeat as potential predictors for EMD. The analyses concerning the impact of hypogonadism were performed both with and without including men on testosterone replacement. Furthermore, these two groups – those treated with testosterone and those not treated - were compared to each other. The CAG length was evaluated as a continuous variable as well as divided into four categories, (<20, 20-21, 22-23 and >23) using the shortest CAG length interval as reference. The GGN repeat length was categorized into three groups, (<23, 23 and >23) using the most common length of 23 as the reference. When assessing the impact of hypogonadism, potential confounders such as: age, smoking, body mass index (BMI) and laboratory (Lund vs. Stockholm) were included in the models, one at a time. These factors were kept within the model if they changed the risk estimate more than 15 %.

The association between the different treatment modalities and EMD was evaluated using Fisher's exact test. One model was applied comparing all treatment groups to each other and one comparing HDCT to all the others. The reason for not using logistic regressions when comparing treatment groups was the low numbers of individuals in some of the treatment modality groups.

Results

Impact of therapy and AR polymorphism on sperm concentration (article I)

Azoospermia

In total 73 men delivered post-orchidectomy samples before other treatment. Among those four (5.5 %, 95 % CI: 1.5-13 %) were azoospermic. One was still azoospermic at six months and one patient did have few sperms in the ejaculate at twelve months. For two patients no post-treatment samples were yet available. Of the 69 men having spermatozoa in the ejaculate after orchidectomy, 16 have not yet delivered any post-treatment samples. Among 53 patients who delivered one or more post-treatment samples, five became azoospermic after CT or RT treatment (9.4 %, 95 % CI: 3.1 -21 %). None of the 26 men in the ACT became azoospermic (95 % CI: 0-13 %). Among seventeen patients in the SCT group, two were azoospermic after six and one at sixty (18 %, 95 % CI: 3.8-43 %) months. From these three patients only one post-treatment sample was available. Two of ten men in the RT group (20 %, 95 % CI: 2.5-56 %) were azoospermic six months after treatment, one regained sperm production after one year, whereas the other did not yet have further follow-up.

Sperm concentration

No significant decrease in sperm concentration was seen, at any time point, in men who received 1 to 2 cycles of adjuvant chemotherapy, ACT group (Figure 6).

Pre-treatment vs. six months

Pre-treatment vs. twelve months



Pre-treatment vs. two-five years



Figure 6: Sperm concentration, longitudinal data: ACT, SCT, RT as defined in the text. Bars correspond to median value, boxes to the interquartile interval and whiskers to 95 % CI. (A) Pre-treatment vs. six months; (B) Pre-treatment vs. twelve months; (C) Pre-treatment vs. two to five years. Black boxes represent pre-treatment values. Adapted from Eberhard et al, Human Reproduction 2004.

In the SCT group, the median sperm concentrations after six and twelve months were significantly lower than before treatment, 0.05 vs. 3.4 x 10^{6} /mL (p=0.043) after six months and 2.4 vs. 18 x 10^{6} /mL (p=0.046) after twelve months corresponding to a decease of 99 % and 87 % respectively. After two to five years the sperm concentration had increased to 19 x 10^{6} /mL, not statistically significantly different from the pretreatment value of 12 x 10^{6} /mL (p=0.8) (Figure 6).

In patients treated with RT, the medium sperm concentration decreased from 48 x 10^6 /mL pre-treatment to 0.1 x 10^6 /mL (p=0.04) at six months corresponding to a decrease of almost 100 %. After twelve months there was a decrease in concentration, 6.3 vs. 36 x 10^6 /mL pre-treatment, however not statistically significant (p=0.075). After two to five years the sperm concentration had increased to 47 x 10^6 /mL, not statistically significantly different from the pretreatment value of 32 x 10^6 /mL (p=0.27) (Figure 6). At twelve month the concentration had increased significantly to 6.8 x 10^6 /mL vs. 0.9 x 10^6 /mL at six months (p=0.03).

There was no significant difference in median sperm concentration before treatment, neither between the NSGCT and SGCT patients, nor between the treatment groups. After six and twelve months there was a statistically significant difference between the groups (p=0.0001), the ACT group having significantly higher sperm concentration than both the SCT (p=0.0001) and the RT group (p=0.001). Concentrations after two to five years did not differ between the therapy groups.

Factors predicting sperm concentration

Type of treatment, but not the pre-treatment sperm concentration or CAG/GGN repeat lengths, independently predicted concentration after six months (p<0.0005) and after one to two years (p=0.004). No statistically

significant association with the treatment modality was found after three to five years.

When analyzing the group of men receiving SCT separately, only the CAG length (p=0.02) but neither GGN number, sperm concentration postorchidectomy nor age did significantly predict the concentration after one to two years.

Correlation between sperm recovery and AR polymorphisms

There was a significant correlation between CAG repeat and sperm concentration after one to two years in men treated with 3 or 4 cycles of CT, SCT, (rho= -0.72; p=0.03) (Figure 7). The repeat number did not correlate with sperm concentration in the other therapy groups or at other time points – including postorchidectomy. No correlation was found for GGN length.



Figure 7: Correlation between CAG repeat length and sperm concentration in patients treated with three to four cycles of chemotherapy one to two years after treatment. Adapted from Eberhard et al, Human Reproduction 2004.

Risk factors for developing hypogonadism (article II)

Post-treatment hypogonadism, relation to hypogonadism before treatment

Hypogonadism postorchidectomy, but before further treatment, strongly predicted hypogonadism after six (OR 53, 95 % CI: 19-145), twelve (OR 125, 95 % CI: 37-430), twenty-four (OR 88, 95 % CI: 26-300) and thirty-six (OR 121, 95 % CI: 32-460) months. Postorchidectomy, 22 of 58 men (38 %) were hypogonadal. After one year, twelve out of twenty (60 %) and after two years six out of eleven (55 %) had normal hormone levels (Figure 8).



Figure 8: Odds ratios for hypogonadism for TGCC patients in relation to hypogonadism at T0. T0=postorchidectomy samples. T6, T12, T24, T36 and T60=six, twelve, twenty-four and thirty-six months after treatment. Adapted from Eberhard et al, Eur. J. Endocrin., 2008.

Hypogonadism in relation to treatment

For the SCT group, a significantly increased OR for hypogonadism, as compared to the ACT group, was found after six (OR 22, 95 % CI: 4.4-118) and twelve (OR 5.8, 95 % CI: 1.5-22) months post therapy (Figure 9).



Figure 9: Therapy dependent odds ratios of developing hypogonadism with the group given adjuvant chemotherapy as reference in patients treated with three to four cycles of chemotherapy. T0=postorchidectomy samples. T6, T12, T24, T36 and T60=six, twelve, twenty-four, thirty-six and sixty months after treatment. Adapted from Eberhard et al, Eur. J. Endocrin., 2008.

Also in the RT group, as compared to the ACT treated men, a significantly increased OR was observed after six (OR 10, 95 % CI: 2.1-47) and after twelve (OR 3.9, 95 % CI: 1.1-14) months post therapy (Figure 10).

Hypogonadism in relation to age, stage and androgen receptor polymorphisms

No statistically significant relation to the risk of hypogonadism was observed for age, stage of disease or androgen receptor CAG and GGN repeat lengths.



Figure 10: Therapy dependent odds ratios of developing hypogonadism with the group given adjuvant chemotherapy as reference in radiotherapy treated patients. T0=postorchidectomy samples. T6, T12, T24, T36 and T60=six, twelve, twenty-four, thirty-six and sixty months after treatment. Adapted from Eberhard et al, Eur. J. Endocrin., 2008.

Hypogonadism in relation to testicular characteristics

Testicular microlithiasis was predictive for hypogonadism postorchidectomy (OR 11, 95 % CI: 1.2-112) and twelve (OR 3.9, 95 % CI: 1.1-13), twenty-four (OR 3.0, 95 % CI: 1.0-8.8), thirty-six (OR 5.4, 95 % CI: 1.7-17) and sixty (OR 4.4, 95 % CI: 1.2-16) months post-treatment (Figure 11). When excluding patients being hypogonadal postorchidectomy a similar trend was observed, reaching statistical significance after thirty-six (OR 5.2, 95 % CI: 1.5-19) and sixty (OR 4.7, 95 % CI: 1.0-21) months. The contralateral testicles volume and consistency were not associated with any increased risk for hypogonadism.



Figure 11: Odds ratios for hypogonadism in relation to microlithiasis of the remaining testicle. T0=postorchidectomy samples. T6, T12, T24, T36 and T60=six, twelve, twenty-four, thirty-six and sixty months after treatment. Adapted from Eberhard et al, Eur. J. Endocrin., 2008.

Sexual function and relation to hypogonadism and treatment (article III)

Sexual function

Patients reported more often manifest low sexual desire (adjusted OR 6.7, 95 % CI: 2.1-21.0), and manifest erectile dysfunction (adjusted OR 3.8, 95 % CI: 1.4-10.0) than the reference group. Twelve percent of TGCC patients reported erectile dysfunction compared to 3 % for controls, whereas the corresponding figures for low sexual desire were 4 % and 2 % respectively. Low sexual desire and manifest erectile dysfunction per se as well as manifest erectile distress concurred for 25 % of the patients and 40 % of the reference group, respectively. No other significant differences between

patients and comparators were found when adjustment for potential confounders was made (Table 3).

Sexual function in relation to hypogonadism

Twenty-nine percent of the men were biochemically hypogonadal. No significant association between biochemical hypogonadism and low sexual desire (OR 1.2, 95 % CI: 0.11-14) or manifest erectile dysfunction (OR 1.1, 95 % CI: 0.26-4.5) was found. Inclusion of patients treated with testosterone did not change the OR estimates given above. Using LH above 10 IU/L as the only indicator of hypogonadism did not change the results.

Sexual function in relation to treatment modality

With a few exceptions, the three therapeutic groups SO, RT and ACT did not differ significantly from the SCT-group regarding the age-adjusted OR's for sexual dysfunctions. The only statistically significant difference between the therapy groups was more prevalent erectile dysfunction (OR 8.8, 95 % CI: 1.2-62) and perception of their sexual life as not satisfying (OR 9.9, 95 % CI: 1.7-58) in SO patients, as compared to the SCT group. Furthermore, a higher proportion of patients in the SCT group than in the SO group reported having intercourse during the past five days (OR 9.9, 95 % CI: 1.7-58). **Table 3:** OR for TGCC patients compared to Swedish nationally representative comparators for variables assessing sexual function, frequency of intercourse, sexual satisfaction and treatment seeking.

I

	Age-adjusted OR (95 % CI)	Adjusted OR* (95 % CI)
Sexual dysfunctions:		
Low sexual desire	2.4 (0.85-6.8)	6.7 ^{acd} (2.1-21)
Less sexual desire than 5 years ago	0.83 (0.56-1.2)	0.83 (0.56-1.2)
Decrease in sexual interest	0.91 (0.53-1.7)	0.91 (0.53-1.7)
Erectile dysfunction ^f	6.6 (3.1-14)	3.8^{bd} (1.4-10)
Erectile dysfunctional distress ^f	6.5 (2.6-16)	2.5 ^{bd} (0.65-9.8)
Premature ejaculation ef	0.90 (0.40-2.0)	0.68 ^{abd} (0.29-1.6)
Delayed ejaculation ^{ef}	3.9 (0.94-15.9)	2.4 ^{bd} (0.45-12.9)
Less than 5 days since last sexual intercourse.	0.71 (0.50-1.1)	0.71 (0.50-1.1)
More than 3 months since last sexual intercourse	1.1 (0.57-2.0)	1.3 ^b (0.67-2.7)
Satisfaction: Dissatisfying sex life	0.84 (0.50-1.4)	0.84 (0.50-1.4)
<i>Treatment seeking:</i> Need for sexual advice or help ^e	0.87 (0.50-1.5)	0.87 (0.50-1.5)
Consulted an expert for sexual advice or help ^e	0.99 (0.84-1.2)	0.99 (0.84-1.2)

*Adjusted for age and one or more of the variables a-e described below. a=occupation b=children/have or want to c=smoking/snuffing d=Body Mass index e= eight patients with absence of antegrade ejaculation are excluded f=nine patients and 69 men from general population not having had intercourse the last twelve months excluded

EMD in relation to hypogonadism, AR polymorphism and treatment (article IV)

Frequencies of emotional disorders

Among the 165 patients, 19 % scored \geq 8 on HADS-A and 5 % on the HADS-D. Among the 20 patients on testosterone replacement therapy, 30 % scored \geq 8 on HADS-A and 10 % on the HADS-D.

Hypogonadism in relation to EMD

Of the 165 included patients, three refused to deliver a blood sample and twenty were on testosterone replacement therapy. Among the remaining 142, 36% were biochemically hypogonadal. If using only $LH \ge 10$ as the criterion, 20% were hypogonadal.

The risks of anxiety (OR 1.0, 95 % CI: 0.40-2.4) and depression (OR 1.1, 95 % CI: 0.20-6.4) were not increased in biochemically hypogonadal TGCC patients. When men on testosterone replacement therapy (n=20) were compared to those not on replacement (n=142), no difference was found.

AR polymorphisms in relation to EMD

Twenty-six percent had CAG repeat length <20, 28 % had 20-21, 22 % had 22-23 and 24 % had >23. Sixteen percent had GGN repeat length <23, 50 % had 23 and 34 had >23.

There was no significant correlation between AR polymorphisms and EMD. The results were similar irrespective of whether we included or excluded the patients on testosterone replacement in the analyses.

EMD in relation to treatment modality

In the SO group, none suffered from EMD. In the ACT group, 10 % suffered from anxiety and 2 % from depression. In the SCT group, 15 % had anxiety and 2 % depression and the corresponding figures in the RT group were 29 % and 10 %, respectively. Finally, in the HDCT group, 62 % had anxiety and 12 % depression (Table 4). When comparing all treatment groups to each other, there was a significant difference regarding frequency of anxiety (p=0.002), but not regarding depression (p=0.19). When the HDCT group, was compared to the other groups, there was a significant difference in the prevalence of anxiety (p=0.006), but not in the prevalence of depression (p=0.38).

	HADS-A \geq 8 Anxiety (n) (%)	HADS-D ≥ 8 Depression (n) (%)				
All patients	31 (19)	8 (4.8)				
Treatment received						
SO n=13	0	0				
ACT n=41	4 (9.8)	1 (2.4)				
SCT n=54	8 (15)	1 (1.9)				
RT n=49	14 (29)	5 (10)				
HDCT n=8	5 (62)	1 (12)				

Table 4. EMD in relation to treatment in 165 TGCC patients 3-5 years after treatment.
Discussion

Major findings and clinical implications

This thesis provided information which may be valuable in the counselling and management of patients treated for TGCC:

(1) Patients receiving adjuvant RT or SCT are at high risk of developing severe oligozoospermia, with high chance of recovery 24 months after completion of the therapy. On the other hand, ACT has only a minor influence on sperm concentration, but the number of patients was too limited to exclude development of azoospermia in some susceptible subjects. These findings do not, however, change the common practice of offering sperm cryopreservation prior to cancer treatment. In the RT and SCT treated men, the azoospermia may be permanent or prolonged, which may seriously reduce the chance of fathering a child.

(2) The association we found between the AR CAG repeat length and regain of sperm concentration one to two years after SCT is to our knowledge the first report of the impact of genetic factors on the recovery of spermatogenesis after cancer therapy. If this finding can be confirmed in a larger material, the CAG length could serve as a potential instrument in determining inter-individual differences in the rapidity of sperm regeneration after cancer therapy. Polymorphisms in other genes related to spermatogenesis may also prove to be valuable in prediction of the susceptibility to the gonadotoxic effect of cancer therapy.

(3) TGCC patients are at increased risk of androgen deficiency, which to a certain degree has not received enough attention in the follow-up of these

men. A significant increase in the OR for hypogonadism was observed six to twelve months post-treatment in patients receiving three to four cycles of chemotherapy and in the RT group, with a decreased risk after two to five years. Previous studies on the risk of hypogonadism in men treated for TGCC (Brennemann et al, 1997; Hansen et al, 1990; Palmieri et al, 1996) did not clearly discriminate between treatment modalities and/or the posttreatment investigations were not performed at specific time-points, which is why a more precise mapping of the process of Leydig cell recovery could not be done. An important observation was not only the increased OR of hypogonadism but also the possible recovery of the Leydig cell function, warranting some hesitation in initiation of androgen replacement. Microlithiasis in the remaining testicle was a risk factor for hypogonadism both pre- and post-treatment. In addition, hypogonadism prior to chemo-or radiotherapy was another predictor of post-treatment hypogonadism, whereas AR polymorphisms and age as well as testicular volume did not have any predictive value. Defining risk factors of hypogonadism enables the identification of patients to be recommended for future screening for androgen deficiency, including testosterone and LH measurements as well as andrological counselling.

(4) Three to five years after cancer treatment, low sexual desire and manifest erectile dysfunction were commoner among TGCC patients than in the age-matched general male population. Interestingly, these outcomes were associated with neither biochemical signs of hypogonadism nor with treatment intensity. From a clinical point of view, erection seemed to be the most important issue, since as many as 12 % of the patients reported frequent erectile dysfunction during the last year compared with only 3 % in the general population. This impairment of sexual function in TGCC survivors should also be given more attention and looking for signs of

sexual dysfunction should not be restricted to hypogonadal and/or heavily treated men.

(5) Even emotional disorders (EMD) seem to be rather frequent in TGCC survivors, with anxiety found in 19 % and depression in 5 %, three to five years after treatment. The risk of these EMD was not associated with markers of androgen action as biochemical signs of hypogonadism or AR polymorphisms. The intensity of treatment influenced the risk of anxiety, being present in 62 % of patients treated with \geq 5 cycles of chemotherapy. Our findings indicate, however, that hypogonadism and emotional disorders, although sharing some symptoms, seem to be two different entities.

Spermatogenesis and Leydig cell function

In agreement with earlier reports, we found the negative impact of CT on spermatogenesis to be dose-dependent (Petersen *et al*, 1994) and ACT did not have any significant influence on sperm production (Cullen *et al*, 1996). The time course of the recovery of sperm production was similar in men treated with RT and SCT, with pre-treatment levels of sperm concentration reached after 24 to 60 months (Figure 6). Also, in agreement with earlier reports, the type of therapy predicted sperm concentration (Brydoy *et al*, 2005; Pont & Albrecht, 1997). We found treatment modality to be a predictor of sperm concentrations after six months, but also after one to two years.

We found no patients developing azoospermia after ACT, but the number of men included in the study was too low to exclude that this side-effect may occur in some of them. Almost 40 % of SCT or RT treated men developed azoospermia six months after treatment. Since follow-up samples were

lacking in most of these men, however, no firm conclusions regarding the subsequent period could be drawn.

With regard to postorchidectomy, before further treatment, 43 % was hypogonadal among patients later treated with SCT or RT. In these patients, analogously with the effects of the treatment on sperm concentration, we found decreased testosterone levels and/or increased LH in 64 %, six to twelve months post-treatment. Corresponding figures after two years were 46 % and after five years 24 % (Figures 9 & 10). The time-related posttreatment recovery of Leydig cell function is in discordance with the data published by Nord and colleagues, who found an increased risk of hypogonadism ten years after completion of TGCC treatment (Nord *et al*, 2003). An age-dependent deterioration of Leydig cell function, related to a longer follow-up period cannot, however, be excluded.

Although stages of disease and treatment intensity are closely related, the latter but not the former was associated with the risk of hypogonadism. Among stage 1 patients, however, some were treated with ACT and some with RT. These two treatment modalities differ in their impact on Leydig cell function, which might explain why disease stage was not a predictor of post-treatment Leydig cell insufficiency.

Scrotal ultrasound is routinely used in the diagnosis of TGCC. Among 107 patients evaluated with ultrasonography, 46 % had microlithiasis in the contralateral testis, which is in concordance with earlier reports (Costabile, 2007). The finding was associated with post-treatment risk of hypogonadism with an OR of 3-11. We found an increased risk of post-treatment hypogonadism after six months and at subsequent time-points in men with microlithiasis. Since microlithiasis has been reported in patients with testicular dysgenesis syndrome (TDS) (26) and is believed to be associated with confirmed testicular cancer (27), pre-treatment Leydig cell

dysfunction might be the cause of post-therapy hypogonadism. Even after exclusion of the subgroup of men who were hypogonadal postorchidectomy, before further therapy, however, microlithiasis remained a predictor of hypogonadism. To my knowledge there are no studies indicating that cancer therapy can induce microlithiasis and therefore I assume that the pathological ultrasonic pattern was also present prior to treatment.

Sexual and psychological function

Impairment of sexual function among TGCC patients, including decrease in sexual desire, ejaculation, orgasm, sexual satisfaction, sexual activity, libido, arousal and erection, have previously been reported by others (Fegg *et al*, 2003; Jonker-Pool *et al*, 1997). These studies did not, however, include control groups and patients with varying post-treatment observation time were included, thus not considering possible dynamic changes in these conditions related to post-treatment stress and/or possible recovery over time. Our study also lacks the longitudinal aspect, but it is valid for the specific three- to five-year follow-up period.

Aas and colleagues (Aass *et al*, 1993) followed 76 patients longitudinally with a questionnaire before and up to 36 months after treatment. They found that cancer treatment had a negative impact on the patient's satisfaction with his sexual life initially after cancer therapy, but this problem partly resolved later in the follow-up. Lackner and colleagues found no increased risk of erectile dysfunction in TGCC survivors, possibly because of the low statistical power of the study (Lackner *et al*, 2005). Also in a case-control study on stage 1 and 2 radiotherapy-treated patients, no difference in frequency of sexual dysfunction between patients and healthy controls was observed (Incrocci *et al*, 2002). The follow-up period, however, varied from

one month to ten years, which, at least partly, might invalidate the conclusions of this study.

Our findings in TGCC patients of a decreased erectile function and low sexual desire might be considered as different ways of measuring the same outcome, as it is known that low sexual desire may cause erectile dysfunction and vice versa (Fugl-Meyer & Fugl-Meyer, 2002). In our study 40 % of the reference group, but only 25 % of the patients, who stated that they had erectile dysfunction, also had low sexual desire. This indicates that in TGCC patients these dysfunctions should be considered, at least partly, as separate conditions.

A low level of sexual satisfaction generally accompanies all sexual dysfunctions and in particular erectile distress (Lewis et al, 2004). The finding that the reference group and the patient group had similar prevalence of not being sexually satisfied, despite the patients' higher prevalence of erectile and desire dysfunctions, suggests that the patients were reasonably psychologically adjusted to their situation. In line with this are the findings that there was no difference in reported erectile distress or in the need for sexual advice between the patients and the reference group. Interestingly, although sexual desire was low among the TGCC subjects, they did not differ in regard to incidence of decrease of sexual desire during the last five years, compared with the general population. These two parameters differ from each other, the first reflecting the present situation, while the latter refers to change over time. This may indicate that the relatively low sexual desire in TGCC patients is not related to the cancer diagnosis or the therapy, a suggestion supported by the fact that no significant difference was seen in sexual desire between the therapy groups. It is, therefore, tempting to hypothesise that low sexual desire in the TGCC men may rather be associated with the disease per se than with its treatment.

Male hypogonadism and EMD share several symptoms including low levels of resolution, ambivalence, impaired concentration, fatigue and lethargy (American Psychiatric Association, 1995; Carnegie, 2004). It could, therefore, be anticipated that disease- and treatment-induced hypogonadism is the underlying cause of EMD in TGCC patients. This could also be hypothesised from a biological point of view, since AR receptors are found in several parts of the brain (Beyenburg *et al*, 2000) and possibly several neurological, cognitive and psychiatric conditions are influenced by androgen action (Almeida *et al*, 2004; Cherrier *et al*, 2005; Colangelo *et al*, 2007). Our results indicate, however, that the EMD symptoms in TGCC patients are not related to hypogonadism. The two conditions seem to be two separate entities which, from a clinical point of view, may represent a problem in making the right diagnosis.

Our results are in accord with those presented by Wiechno and colleagues (Wiechno et al, 2007), who also reported no association between the results of the HADS and elevated LH or low testosterone. When using another depression scale, however, the Beck Depression Inventory (BDI), they noted an increased risk of depression when LH levels were increased. Both scales are well validated and the sensitivity and specificity in detecting major depression is very high for both. In the case of minor depression, however, there are indications that in cancer patients following curative treatment (Katz et al, 2004), the HADS demonstrates greater specificity, sensitivity and also a higher positive predictive value, although the difference between the scales was found not to be statistically significant. Although the study was based on a limited number of patients and therefore needs confirmation from a larger sample, the finding of a 62 % risk of having anxiety three to five years after treatment in the HDCT group raises both clinical concerns and aetiological speculations. In the clinical followup, these patients should be evaluated for anxiety disorders. The aetiology

could be because of the psychological stress, these young patients experiencing a serious threat to their lives. This EMD could also be related, however, to the high doses of CT. In a recent report, no cognitive impairment was seen in TGCC patients receiving CT compared with orchidectomised +\- RT patients. In this study, however, the CT patients received standard treatment doses (Pedersen et al, 2009). There are recent data indicating a dose-dependent cognitive dysfunction following cisplatinbased CT (Skoogh, 2008), which is also reported for chemotherapy-treated patients with other diagnosis (Nelson et al, 2007). Current research indicates that the cognitive domains that may be most affected by chemotherapeutic agents are visual and verbal memory, language, attention, and psychomotor functioning and subsequent anxiety. The potential mechanisms that cause such disruption remain largely unknown, although contributing factors could be vascular injury and oxidative damage, inflammation, direct injury to neurons or autoimmune responses (Nelson et al, 2007; Skoogh, 2008). Another study reported negative impact of CT in breast cancer patients on selected domains of cognitive function. These changes remained significant even after controlling for anxiety, depression, fatigue and haemoglobin level, which might support a psychological mechanism for anxiety among HDCT patients (Jansen et al, 2008).

Impact of androgen receptor polymorphism

We found a negative correlation between the length of CAG repeat in the AR gene and sperm concentration one to two years post three to four cycles of BEP. Furthermore, in a multivariate analysis, together with sperm concentration postorchidectomy, the CAG length was shown to be a significant, independent, predictor of sperm concentration in SCT-treated patients after one to two years. Previous in vitro and in vivo studies have

shown that the length of the CAG repeat is inversely correlated to the transcriptional activity of the AR and thereby to the sensitivity to the androgens (Tut *et al*, 1997). Some studies have demonstrated longer CAG repeats in infertile men (Dowsing *et al*, 1999) and an inverse correlation between sperm concentration and CAG length (von Eckardstein *et al*, 2001a). We did not find any correlation between CAG lengths and pre-treatment sperm concentration. Furthermore, the results of the multivariate analysis indicated that the effect of this AR polymorphism is not exerted through regulation of the pre-treatment state of spermatogenesis, but is rather implicated in the process of recovery. We did not find any association between sperm number and the length of the other repetitive sequence of the AR gene – the GGN repeat.

The finding of an association between the CAG segment and recovery of spermatogenesis is intriguing. Since androgens are mostly involved in the regulation of post-meiotic stages of spermatogenesis (Sofikitis *et al*, 2008), our finding indicates that after SCT recovery of late stages of spermatogenesis (Zhang *et al*, 2003) plays an important role in reaching pre-treatment levels of sperm concentration. Our study indicates that decreased androgen action after BEP treatment delays the recovery of sperm production. Inclusion of larger groups of men is necessary, however, to draw any firm conclusions regarding this issue.

Lower androgen sensitivity, with higher LH levels, was found in men with long CAG tracts (28). It could be anticipated that the risk of hypogonadism is modified by the length of this repeat. We found no association, however, between any of the AR polymorphisms and biochemically-defined hypogonadism.

In elderly Finnish men, long CAG repeats were found to be associated with the risk of depression (Harkonen *et al*, 2003) indicating a more important role of androgens in the pathogenesis of EMD in this category of subjects, while in an investigation on adolescent men, short CAG repeats pointed towards not a higher incidence but more severe depressive symptoms (Su *et al*, 2007). Additionally, higher CAG number was associated with lower scores on cognitive tests, maybe also implying an association to EMD (Yaffe *et al*, 2003). Our study did not give any support for an impact of AR polymorphisms on EMD among TGCC patients.

Strengths and weaknesses of thesis

Different analytic strategies were applied in the studies which formed part of this thesis, with a longitudinal approach in articles I and II and crosssection analyses in articles III and IV. The longitudinal approach implies a better option for studying causality and also gives a mapping of time-related changes. The disadvantage, however, is a relatively low number of subjects on whom such analyses could be based, and additional studies are needed to prove or disprove our findings. For the two latter articles, longitudinal analyses will be possible when the vast majority of the patients have passed the five-year follow-up, which will take another one or two years. Both articles have, however, been based on well-validated instruments including a considerable number of subjects. In article III the questions used to evaluate sexual function and satisfaction were validated in a large population study, which also provided data for an age-matched reference group.

The participation rate in the two first studies was quite high, 82 %, reducing the risk of selection bias. In articles III and IV there were two inclusion criteria, primarily to be included in the fertility study and secondarily to have passed the three- to five-year control. The participation rate in these articles was 79 %.

In the third study, the 59 % response rate for the controls may appear somewhat low, but post hoc analyses (Fugl-Meyer *et al*, 2000) have shown that the studied male population is adequately representative of Swedish men aged eighteen to 74. Seventy-five of the 409 (461 eligible minus 52 excluded) men (18 %) who were asked to participate in the fertility study declined to take part. In comparison with those 129 who were finally included in study III (Figure 5), we found no difference regarding their age, but the men not willing to participate had received significantly less advanced treatment. Since we did not find any obvious impact of the treatment intensity on the risk of sexual dysfunction, however, we do not think that this would influence the results.

In study I information about abstinence time was only available for 91 of 177 samples (51 %). It could be anticipated that semen samples collected postorchidectomy, but before further treatment, compared with other time-points, were preceded by a longer abstinence time owing to the disease as well as surgery-related stress. No significant difference, however, in abstinence time was found between samples collected at different time-points.

The signs of hypogonadism are non-characteristic and good biochemical markers of androgen deficiency are lacking. The clinical diagnosis is usually based on the combination of symptoms and serum levels of testosterone and LH. In our studies we have only used biochemical parameters in defining hypogonadism. The levels of 10 nmol/L for total testosterone and 10 IU/L for LH are generally accepted as useful markers of hypogonadism in younger males (22). A shortcoming in articles II, III and IV is the change in the methods for testosterone and LH measurements. In the material from Lund, however, conversion of the results from the method used in the first part of these studies to the one applied during the second part was based on more than 30 subjects tested with both methods.

Furthermore, no statistically significant difference between measurements performed with the different methods was found at any time-point. In the material from Stockholm, no methodological changes were made during the interval from study initiation to the time of data analysis. The reference intervals used in the Lund material after conversions were identical to those used in Stockholm. Finally, the type of method applied for hormone measurement was included as a confounding factor in the statistical analysis in studies II to IV. We therefore believe that our results are reliable despite this methodological drawback.

For logistic reasons, we obtained blood samples at different time-points between 9 a.m. and 3 p.m. This might have blurred the difference between truly hypogonadal and eugonadal men, owing to the diurnal variation of testosterone. It has been reported, however, that the diurnal variation in testosterone levels is less pronounced in hypogonadal men (Winters, 1991). Furthermore, the association between hypogonadism and the outcomes in papers II to IV was unchanged when LH above 10 IU/L was used as the only indicator of androgen deficiency. Unlike testosterone, LH does not decrease during the day. Studies regarding the association between testosterone levels and metabolic signs of hypogonadism reported no difference in the risk estimates regardless of whether the time of blood sampling was taken into consideration or not (Agledahl et al, 2008). In study II, men on androgen replacement were considered hypogonadal from the time of initiation of therapy. In studies III and IV, men already on androgen replacement therapy were excluded, which could lead to an underestimation of the prevalence of patients with problems related to hypogonadism and TGCC. Inclusion of patients on testosterone replacement, however, did not influence the magnitude of the risk estimates in either of these two studies.

General conclusions

- Adjuvant chemotherapy did not induce a significant decline in sperm concentration. After three to four cycles of chemotherapy and adjuvant radiotherapy against abdominal lymph nodes a reduction of sperm concentration was observed, recovering to pre-treatment levels two to five years post-treatment.
- In patients treated with three to four cycles of chemotherapy, androgen receptor CAG number was associated with the recovery of spermatogenesis.
- Hypogonadism postorchidectomy, but before further treatment and testicular microlithiasis, and type of treatment were predictive factors for the risk of post-treatment hypogonadism.
- Compared with the general age-matched population, TGCC patients three to five years after completion of therapy were at significantly higher risk of having low sexual desire and erectile dysfunction. These sexual dysfunctions, however, were not significantly associated with treatment intensity or hypogonadism.
- Biochemical hypogonadism and androgen receptor polymorphism seem not to be risk factors for anxiety or depression in TGCC patients.
- Patients with refractory or relapsed disease receiving five or more cycles of cisplatinum-based chemotherapy may, to a higher degree than patients receiving less intense therapy, suffer from anxiety.

Future Perspectives

The material from the study provides an excellent source to answer several future questions of interest and importance, mainly from a clinical, but also from a biological, perspective. It can provide further data for subgroups of TGCC patients, but also improve our knowledge on inter-individual differences among these patients. Such knowledge can be of value for more individualised counselling concerning several treatment-related side-effects, such as need for cryopreservation, susceptibility to developing infertility, hypogonadism and possibly also sexual and affective disorders. The possibility of evaluating the material with a longitudinal design constantly increases, as the collection of data is ongoing until definite study closure June 2011.

Concerning sperm parameters, a conformational longitudinal study is warranted, including more semen samples with parallel analyses of FSH and inhibin B. The aim would be to evaluate further the risk of developing azoospermia and to investigate the predictive value of the hormone values in relation to assessment of spermatogenesis and its recovery. The finding that in patients treated with three to four cycles of chemotherapy the androgen receptor CAG number was associated with the recovery of spermatogenesis is intriguing, but since it is based on nine subjects only, it definitely needs confirmation in a larger sample. Other polymorphisms in strategic genes should also be studied.

More information regarding the Leydig cell function can be obtained by using the hCG test data, with serum levels of testosterone measured before and 96 hours after hCG administration. In the present study, the hCG test has been performed one and five years after completion of therapy, which

may reveal some more discrete impairments of Leydig cell function, not apparent when measuring testosterone levels without any stimulation (data not yet analysed).

In parallel with current projects, another project has been focusing on sperm DNA integrity in the same patient material (Stahl *et al*, 2004; Stahl *et al*, 2006). Possible associations between defect sperm DNA and genetic polymorphisms and Leydig cell dysfunction, as well as testicular microlithiasis, could also be investigated

The questionnaire includes a well-validated instrument for measuring general satisfaction, the LiSat-11 checklist (Fugl-Meyer *et al*, 2002) also included in the national survey. These parameters should be evaluated longitudinally to reveal causality and association with hypogonadism, sexual dysfunctions and treatment intensity as well as socio-demographics, in order to find potential agents related to low general satisfaction.

Both sexual dysfunctions and emotional disorders should be further evaluated with a longitudinal approach. Some findings need confirmation in a larger sample, in particular the increased risk of anxiety in HDCT patients. Furthermore, the risk of developing depression depending on treatment modality was not possible to evaluate in the current thesis owing to the insufficient number of depressed patients in the subgroups.

Metabolic syndrome is overrepresented in hypogonadal men and also among TGCC patients and the causes are still largely unknown. A direct link has been hypothesised and in that context the study could be expanded with data concerning metabolic syndrome and outcomes tested for correlations to treatment intensity, hypogonadism and AR polymorphisms. The aim would be to find risk factors for developing metabolic syndrome.

The finding of a predictive value of microlithiasis in relation to the risk of hypogonadism both pre- and post-treatment should also be expanded to an investigation of possible association between this ultrasound pattern and semen quality in TGCC patients. A positive finding would be a support for the TDS hypothesis and also indicate the possibility of using ultrasonographic investigation as a tool in prediction of future fertility.

Acknowledgement

Many people have been involved in the production of this work. There is one person who has been indispensable for almost all parts of this thesis, including ideas, statistics, writing, conclusions as well as intellectual input. Add to this a never ending enthusiasm and encouragement as well as intellectual and also economic support. Without him nothing would have been made by this pen. I hope that his efforts in trying to develop my scientific mind will bear future fruits. Therefore I especially want to thank my brilliant supervisor and mentor, **Aleksander Giwercman**. Furthermore, during my time under his wing, besides the obvious and expected difficulties with the research, I also experienced a very difficult time in my private life. His support, tolerance and empathy during this period were of extra value.

I also want to give special thanks to my co-supervisors, **Eva Cavallin-Ståhl** and **Yvonne Giwercman**. They have in a splendid way supplemented Aleksander: Eva, with an enormous knowledge about oncology and a huge experience in sciences; Yvonne with great knowledge in the field of microbiology, but also an effective examiner of scientific content. Another person, crucial for this work is **Olof Ståhl**. Together we have included most of the patients in Lund and we have collected, discussed and evaluated our results. Not only being a perfect colleague, he is also a very close friend. **Gabriella Cohn-Cedermark** started the inclusion in Stockholm and thanks to her, we can at definite study closure in 2011, look forward to a huge longitudinal material. In addition, she has also been important as a guarantor of the material collection in Stockholm.

Malin Eberhard-Gran, also being my beloved sister, has been important for the epidemiology and psychiatry, Eva-Cecilia Salmonson for the ultrasonography, Kerstin Fugl-Meyer for the sexology and Lars Rylander for the statistics: all of them making this thesis a lot better.

Per Flodgren and **Magdalena Cwikiel** at the Department of Oncology, Lund and **Katinka Sandberg** and **Agneta Richard-Holm** at South Hospital, Karolinska University Hospital helped with the inclusion of patients. **Hamideh Rastkhani, Kristina Lundin** and **Camilla Anderberg** made the DNA analyses for the androgen receptor polymorphisms. **Annette Möller** helped with organizing the patient inclusion procedure and for administrating the questionnaires in Lund and **Ulrik Kvist, Eva Lilliehöök** and **Emelie Ekwurtzel** for patient flow and sampling in Stockholm. I also want to thank **Carsten Rose**, my clinical employer and **Anders Johnsson**, my clinical supervisor. Both of them have supported me with lots of time necessary to accomplish this work.

I also appreciate the research group at **CRC**, **Malmö**, colleagues at the **Department of Oncology, Lund** and the staff at **RMC**, **Malmö** for all being stimulating in different ways.

The study was supported by grants from Swedish Government Funding for Clinical Research, the Swedish Cancer Society, Region Skåne, Gunnar Nilssons Cancerstiftelse, Swedish Childhood Cancer, Malmö University Hospital Foundation for Cancer Research and Foundation for Urological Research and King Gustaf V's Jubilee fund for Cancer Research, Stockholm. I also, in addition to my sister **Malin**, want to thank my brothers, **David**, **Mårten** and **Jonas** for all the discussions through-out time and their unprejudiced support when life is rough, thereby largely being responsible for me being me.

My parents, Marie-Louise and Göran, have always supported me 100 % in all situations. I could not have better parents.

Monica, you are miraculous. Thank you for all your love and support.

Finally, Hugo, Felix and Maja, you are my universe.

Reference list

Aass N, Fossa SD, Theodorsen L, Norman N (1991) Prediction of long-term gonadal toxicity after standard treatment for testicular cancer. *Eur J Cancer* **27:** 1087-91

Aass N, Grunfeld B, Kaalhus O, Fossa SD (1993) Pre- and post-treatment sexual life in testicular cancer patients: a descriptive investigation. *Br J Cancer* **67:** 1113-7

Adami HO, Bergstrom R, Mohner M, Zatonski W, Storm H, Ekbom A, Tretli S, Teppo L, Ziegler H, Rahu M, et al. (1994) Testicular cancer in nine northern European countries. *Int J Cancer* **59:** 33-8

Agledahl I, Skjaerpe PA, Hansen JB, Svartberg J (2008) Low serum testosterone in men is inversely associated with non-fasting serum triglycerides: the Tromso study. *Nutr Metab Cardiovasc Dis* **18:** 256-62

Albers P, Albrecht W, Algaba F, Bokemeyer C, Cohn-Cedermark G, Horwich A, Klepp O, Laguna MP, Pizzocaro G (2005) Guidelines on testicular cancer. *Eur Urol* **48**: 885-94

Almeida OP, Waterreus A, Spry N, Flicker L, Martins RN (2004) One year follow-up study of the association between chemical castration, sex hormones, beta-amyloid, memory and depression in men. *Psychoneuroendocrinology* **29:** 1071-81

American Psychiatric Association (1995) *Diagnostic and statistical manual of mental disorders*, 4. ed edn. Washington, DC: American Psychiatric Press

Anserini P, Chiodi S, Spinelli S, Costa M, Conte N, Copello F, Bacigalupo A (2002) Semen analysis following allogeneic bone marrow transplantation. Additional data for evidence-based counselling. *Bone Marrow Transplant* **30:** 447-51

Axelsson J, Ingre M, Akerstedt T, Holmback U (2005) Effects of acutely displaced sleep on testosterone. *J Clin Endocrinol Metab* **90**: 4530-5

Belchetz PE, Plant TM, Nakai Y, Keogh EJ, Knobil E (1978) Hypophysial responses to continuous and intermittent delivery of hypopthalamic gonadotropin-releasing hormone. *Science* **202:** 631-3

Beyenburg S, Watzka M, Clusmann H, Blumcke I, Bidlingmaier F, Elger CE, Stoffel-Wagner B (2000) Androgen receptor mRNA expression in the human hippocampus. *Neurosci Lett* **294:** 25-8

Bjelland I, Dahl AA, Haug TT, Neckelmann D (2002) The validity of the Hospital Anxiety and Depression Scale. An updated literature review. *J Psychosom Res* **52:** 69-77

Boisen KA, Kaleva M, Main KM, Virtanen HE, Haavisto AM, Schmidt IM, Chellakooty M, Damgaard IN, Mau C, Reunanen M, Skakkebaek NE, Toppari J (2004) Difference in prevalence of congenital cryptorchidism in infants between two Nordic countries. *Lancet* **363**: 1264-9

Bokemeyer C, Berger CC, Kuczyk MA, Schmoll HJ (1996) Evaluation of long-term toxicity after chemotherapy for testicular cancer. *J Clin Oncol* **14**: 2923-32

Bosl GJ, Leitner SP, Atlas SA, Sealey JE, Preibisz JJ, Scheiner E (1986) Increased plasma renin and aldosterone in patients treated with cisplatinbased chemotherapy for metastatic germ-cell tumors. *J Clin Oncol* **4:** 1684-9

Brennemann W, Stoffel-Wagner B, Helmers A, Mezger J, Jager N, Klingmuller D (1997) Gonadal function of patients treated with cisplatin based chemotherapy for germ cell cancer. *J Urol* **158:** 844-50

Brydoy M, Fossa SD, Klepp O, Bremnes RM, Wist EA, Wentzel-Larsen T, Dahl O (2005) Paternity following treatment for testicular cancer. *J Natl Cancer Inst* **97:** 1580-8

Caffo O, Amichetti M (1999) Evaluation of sexual life after orchidectomy followed by radiotherapy for early-stage seminoma of the testis. *BJU Int* **83**: 462-8

Campbell MF, Walsh PC (1997) *Campbell's urology*, 7th edn. Philadelphia: Saunders Co.

Carnegie C (2004) Diagnosis of hypogonadism: clinical assessments and laboratory tests. *Rev Urol* **6 Suppl 6:** S3-8

Centola GM, Keller JW, Henzler M, Rubin P (1994) Effect of low-dose testicular irradiation on sperm count and fertility in patients with testicular seminoma. *J Androl* **15:** 608-13

Cepoiu M, McCusker J, Cole MG, Sewitch M, Belzile E, Ciampi A (2008) Recognition of depression by non-psychiatric physicians--a systematic literature review and meta-analysis. *J Gen Intern Med* **23**: 25-36

Cherrier MM, Matsumoto AM, Amory JK, Asthana S, Bremner W, Peskind ER, Raskind MA, Craft S (2005) Testosterone improves spatial memory in men with Alzheimer disease and mild cognitive impairment. *Neurology* **64**: 2063-8

Colangelo LA, Sharp L, Kopp P, Scholtens D, Chiu BC, Liu K, Gapstur SM (2007) Total testosterone, androgen receptor polymorphism, and depressive symptoms in young black and white men: the CARDIA Male Hormone Study. *Psychoneuroendocrinology* **32**: 951-8

Cooper DE, L'Esperance J O, Christman MS, Auge BK (2008) Testis cancer: a 20-year epidemiological review of the experience at a regional military medical facility. *J Urol* **180:** 577-81; discussion 581-2

Costabile RA (2007) How worrisome is testicular microlithiasis? *Curr Opin Urol* **17:** 419-23

Cullen MH, Stenning SP, Parkinson MC, Fossa SD, Kaye SB, Horwich AH, Harland SJ, Williams MV, Jakes R (1996) Short-course adjuvant chemotherapy in high-risk stage I nonseminomatous germ cell tumors of the testis: a Medical Research Council report. *J Clin Oncol* **14**: 1106-13

Dahl AA, Haaland CF, Mykletun A, Bremnes R, Dahl O, Klepp O, Wist E, Fossa SD (2005a) Study of anxiety disorder and depression in long-term survivors of testicular cancer. *J Clin Oncol* **23:** 2389-95

Dahl AA, Mykletun A, Fossa SD (2005b) Quality of life in survivors of testicular cancer. *Urol Oncol* **23**: 193-200

Davis-Dao CA, Tuazon ED, Sokol RZ, Cortessis VK (2007) Male infertility and variation in CAG repeat length in the androgen receptor gene: a metaanalysis. *J Clin Endocrinol Metab* **92:** 4319-26

De Gendt K, Swinnen JV, Saunders PT, Schoonjans L, Dewerchin M, Devos A, Tan K, Atanassova N, Claessens F, Lecureuil C, Heyns W, Carmeliet P, Guillou F, Sharpe RM, Verhoeven G (2004) A Sertoli cellselective knockout of the androgen receptor causes spermatogenic arrest in meiosis. *Proc Natl Acad Sci U S A* **101:** 1327-32 de Kretser DM (1997) Male infertility. Lancet 349: 787-90

Dearnaley D, Huddart R, Horwich A (2001) Regular review: Managing testicular cancer. *BMJ* **322:** 1583-8

Dieckmann KP, Loy V (1996) Prevalence of contralateral testicular intraepithelial neoplasia in patients with testicular germ cell neoplasms. *J Clin Oncol* **14:** 3126-32

Dieckmann KP, Pichlmeier U (2004) Clinical epidemiology of testicular germ cell tumors. *World J Urol* **22:** 2-14

Donohue JP (2003) Evolution of retroperitoneal lymphadenectomy (RPLND) in the management of non-seminomatous testicular cancer (NSGCT). *Urol Oncol* **21:** 129-32

Dowsing AT, Yong EL, Clark M, McLachlan RI, de Kretser DM, Trounson AO (1999) Linkage between male infertility and trinucleotide repeat expansion in the androgen-receptor gene. *Lancet* **354:** 640-3

Dunn JF, Nisula BC, Rodbard D (1981) Transport of steroid hormones: binding of 21 endogenous steroids to both testosterone-binding globulin and corticosteroid-binding globulin in human plasma. *J Clin Endocrinol Metab* **53:** 58-68

Fegg MJ, Gerl A, Vollmer TC, Gruber U, Jost C, Meiler S, Hiddemann W (2003) Subjective quality of life and sexual functioning after germ-cell tumour therapy. *Br J Cancer* **89:** 2202-6

Fleer J, Hoekstra HJ, Sleijfer DT, Hoekstra-Weebers JE (2004) Quality of life of survivors of testicular germ cell cancer: a review of the literature. *Support Care Cancer* **12:** 476-86

Fossa SD, Abyholm T, Aakvaag A (1984) Spermatogenesis and hormonal status after orchiectomy for cancer and before supplementary treatment. *Eur Urol* **10**: 173-7

Fossa SD, Dahl AA, Loge JH (2003) Fatigue, anxiety, and depression in long-term survivors of testicular cancer. *J Clin Oncol* **21**: 1249-54

Fossa SD, De Angelis P, Kraggerud SM, Evenson D, Theodorsen L, Clausen OP (1997) Prediction of posttreatment spermatogenesis in patients with testicular cancer by flow cytometric sperm chromatin structure assay. *Cytometry* **30:** 192-6

Fossa SD, Lehne G, Heimdal K, Theodorsen L (1995) Clinical and biochemical long-term toxicity after postoperative cisplatin-based chemotherapy in patients with low-stage testicular cancer. *Oncology* **52**: 300-5

Fossa SD, Ous S, Abyholm T, Loeb M (1985a) Post-treatment fertility in patients with testicular cancer. I. Influence of retroperitoneal lymph node dissection on ejaculatory potency. *Br J Urol* **57:** 204-9

Fossa SD, Ous S, Abyholm T, Norman N, Loeb M (1985b) Post-treatment fertility in patients with testicular cancer. II. Influence of cis-platin-based combination chemotherapy and of retroperitoneal surgery on hormone and sperm cell production. *Br J Urol* **57**: 210-4

Fossa SD, Ous S, Lien HH, Stenwig AE (1989) Post-chemotherapy lymph node histology in radiologically normal patients with metastatic nonseminomatous testicular cancer. *J Urol* **141:** 557-9

Fossa SD, Theodorsen L, Norman N, Aabyholm T (1990) Recovery of impaired pretreatment spermatogenesis in testicular cancer. *Fertil Steril* **54**: 493-6

Fry RC, Svensson JP, Valiathan C, Wang E, Hogan BJ, Bhattacharya S, Bugni JM, Whittaker CA, Samson LD (2008) Genomic predictors of interindividual differences in response to DNA damaging agents. *Genes Dev* **22:** 2621-6

Fugl-Meyer AR, Melin R, Fugl-Meyer KS (2002) Life satisfaction in 18- to 64-year-old Swedes: in relation to gender, age, partner and immigrant status. *J Rehabil Med* **34:** 239-46

Fugl-Meyer K, Fugl-Meyer AR (2002) Sexual disabilities are not singularities. *Int J Impot Res* **14:** 487-93

Fugl-Meyer K, Lewin B, Folkhälsoinstitutet (2000) *Sex in Sweden : on the Swedish sexual life 1996*, 1. edn. Stockholm: National Institute of Public Health (Folkhälsoinstitutet)

Fung MM, Bettencourt R, Barrett-Connor E (2004) Heart disease risk factors predict erectile dysfunction 25 years later: the Rancho Bernardo Study. *J Am Coll Cardiol* **43:** 1405-11

Gandini L, Sgro P, Lombardo F, Paoli D, Culasso F, Toselli L, Tsamatropoulos P, Lenzi A (2006) Effect of chemo- or radiotherapy on sperm parameters of testicular cancer patients. *Hum Reprod* **21**: 2882-9

Gascon F, Valle M, Martos R, Ruz FJ, Rios R, Montilla P, Canete R (2000) Sex hormone-binding globulin as a marker for hyperinsulinemia and/or insulin resistance in obese children. *Eur J Endocrinol* **143:** 85-9

Giwercman A, Berthelsen JG, Muller J, von der Maase H, Skakkebaek NE (1987) Screening for carcinoma-in-situ of the testis. *Int J Androl* **10:** 173-80

Giwercman A, von der Maase H, Berthelsen JG, Rorth M, Bertelsen A, Skakkebaek NE (1991) Localized irradiation of testes with carcinoma in situ: effects on Leydig cell function and eradication of malignant germ cells in 20 patients. *J Clin Endocrinol Metab* **73:** 596-603

Giwercman YL, Xu C, Arver S, Pousette A, Reneland R (1998) No association between the androgen receptor gene CAG repeat and impaired sperm production in Swedish men. *Clin Genet* **54**: 435-6

Halvorsen JG, Metz ME (1992a) Sexual dysfunction, Part I: Classification, etiology, and pathogenesis. *J Am Board Fam Pract* **5:** 51-61

Halvorsen JG, Metz ME (1992b) Sexual dysfunction, Part II: Diagnosis, management, and prognosis. *J Am Board Fam Pract* **5**: 177-92

Hansen PV, Hansen SW (1993) Gonadal function in men with testicular germ cell cancer: the influence of cisplatin-based chemotherapy. *Eur Urol* **23:** 153-6

Hansen PV, Trykker H, Helkjoer PE, Andersen J (1989) Testicular function in patients with testicular cancer treated with orchiectomy alone or orchiectomy plus cisplatin-based chemotherapy. *J Natl Cancer Inst* **81**: 1246-50

Hansen SW, Berthelsen JG, von der Maase H (1990) Long-term fertility and Leydig cell function in patients treated for germ cell cancer with cisplatin, vinblastine, and bleomycin versus surveillance. *J Clin Oncol* **8:** 1695-8 Harkonen K, Huhtaniemi I, Makinen J, Hubler D, Irjala K, Koskenvuo M, Oettel M, Raitakari O, Saad F, Pollanen P (2003) The polymorphic androgen receptor gene CAG repeat, pituitary-testicular function and andropausal symptoms in ageing men. *Int J Androl* **26:** 187-94

Hartmann JT, Albrecht C, Schmoll HJ, Kuczyk MA, Kollmannsberger C, Bokemeyer C (1999) Long-term effects on sexual function and fertility after treatment of testicular cancer. *Br J Cancer* **80**: 801-7

Hayes FJ, DeCruz S, Seminara SB, Boepple PA, Crowley WF, Jr. (2001) Differential regulation of gonadotropin secretion by testosterone in the human male: absence of a negative feedback effect of testosterone on follicle-stimulating hormone secretion. *J Clin Endocrinol Metab* **86**: 53-8

Hendry WF, Stedronska J, Jones CR, Blackmore CA, Barrett A, Peckham MJ (1983) Semen analysis in testicular cancer and Hodgkin's disease: preand post-treatment findings and implications for cryopreservation. *Br J Urol* **55**: 769-73

Hoei-Hansen CE, Rajpert-De Meyts E, Daugaard G, Skakkebaek NE (2005) Carcinoma in situ testis, the progenitor of testicular germ cell tumours: a clinical review. *Ann Oncol* **16:** 863-8

Horwich A, Brada M, Nicholls J, Jay G, Hendry WF, Dearnaley D, Peckham MJ (1989) Intensive induction chemotherapy for poor risk nonseminomatous germ cell tumours. *Eur J Cancer Clin Oncol* **25:** 177-84

Howell SJ, Shalet SM (2005) Spermatogenesis after cancer treatment: damage and recovery. *J Natl Cancer Inst Monogr*: 12-7

Huddart RA, Norman A, Moynihan C, Horwich A, Parker C, Nicholls E, Dearnaley DP (2005) Fertility, gonadal and sexual function in survivors of testicular cancer. *Br J Cancer* **93:** 200-7

Huddart RA, Norman A, Shahidi M, Horwich A, Coward D, Nicholls J, Dearnaley DP (2003) Cardiovascular disease as a long-term complication of treatment for testicular cancer. *J Clin Oncol* **21:** 1513-23

Huhtaniemi IT, Pye SR, Limer KL, Thomson W, O'Neill TW, Platt H, Payne D, John SL, Jiang M, Boonen S, Borghs H, Vanderschueren D, Adams JE, Ward KA, Bartfai G, Casanueva F, Finn JD, Forti G, Giwercman A, Han TS, Kula K, Lean ME, Pendleton N, Punab M, Silman AJ, Wu FC (2009) Increased estrogen rather than decreased androgen action is associated with longer androgen receptor CAG repeats. *J Clin Endocrinol Metab* 94: 277-84

Huyghe E, Matsuda T, Thonneau P (2003) Increasing incidence of testicular cancer worldwide: a review. *J Urol* **170:** 5-11

Incrocci L, Hop WC, Wijnmaalen A, Slob AK (2002) Treatment outcome, body image, and sexual functioning after orchiectomy and radiotherapy for Stage I-II testicular seminoma. *Int J Radiat Oncol Biol Phys* **53**: 1165-73

Jacobsen KD, Ous S, Waehre H, Trasti H, Stenwig AE, Lien HH, Aass N, Fossa SD (1999) Ejaculation in testicular cancer patients after postchemotherapy retroperitoneal lymph node dissection. *Br J Cancer* **80**: 249-55

Jansen CE, Dodd MJ, Miaskowski CA, Dowling GA, Kramer J (2008) Preliminary results of a longitudinal study of changes in cognitive function in breast cancer patients undergoing chemotherapy with doxorubicin and cyclophosphamide. *Psychooncology* **17**: 1189-95

Joly F, Heron JF, Kalusinski L, Bottet P, Brune D, Allouache N, Mace-Lesec'h J, Couette JE, Peny J, Henry-Amar M (2002) Quality of life in longterm survivors of testicular cancer: a population-based case-control study. *J Clin Oncol* **20:** 73-80

Jonker-Pool G, van Basten JP, Hoekstra HJ, van Driel MF, Sleijfer DT, Koops HS, van de Wiel HB (1997) Sexual functioning after treatment for testicular cancer: comparison of treatment modalities. *Cancer* **80**: 454-64

Jonker-Pool G, Van de Wiel HB, Hoekstra HJ, Sleijfer DT, Van Driel MF, Van Basten JP, Schraffordt Koops HS (2001) Sexual functioning after treatment for testicular cancer-review and meta-analysis of 36 empirical studies between 1975-2000. *Arch Sex Behav* **30:** 55-74

Joos H, Sedlmayer F, Gomahr A, Rahim HB, Frick J, Kogelnik HD, Rettenbacher L (1997) Endocrine profiles after radiotherapy in stage I seminoma: impact of two different radiation treatment modalities. *Radiother Oncol* **43:** 159-62 Katz MR, Kopek N, Waldron J, Devins GM, Tomlinson G (2004) Screening for depression in head and neck cancer. *Psychooncology* **13**: 269-80

Kenny AM, Raisz LG (2002) Mechanisms of bone remodeling: implications for clinical practice. *J Reprod Med* **47:** 63-70

Kessler RC, Chiu WT, Demler O, Merikangas KR, Walters EE (2005) Prevalence, severity, and comorbidity of 12-month DSM-IV disorders in the National Comorbidity Survey Replication. *Arch Gen Psychiatry* **62:** 617-27

Kessler RC, McGonagle KA, Zhao S, Nelson CB, Hughes M, Eshleman S, Wittchen HU, Kendler KS (1994) Lifetime and 12-month prevalence of DSM-III-R psychiatric disorders in the United States. Results from the National Comorbidity Survey. *Arch Gen Psychiatry* **51**: 8-19

Kittles RA, Young D, Weinrich S, Hudson J, Argyropoulos G, Ukoli F, Adams-Campbell L, Dunston GM (2001) Extent of linkage disequilibrium between the androgen receptor gene CAG and GGC repeats in human populations: implications for prostate cancer risk. *Hum Genet* **109**: 253-61

Klepp O, Dahl O, Flodgren P, Stierner U, Olsson AM, Oldbring J, Nilsson S, Daehlin L, Tornblom M, Smaland R, Starkhammar H, Abramsson L, Wist E, Raabe N, Edekling T, Cavallin-Stahl E (1997) Risk-adapted treatment of clinical stage 1 non-seminoma testis cancer. *Eur J Cancer* **33**: 1038-44

Klepp O, Flodgren P, Maartman-Moe H, Lindholm CE, Unsgaard B, Teigum H, Fossa SD, Paus E (1990a) Early clinical stages (CS1, CS1Mk+ and CS2A) of non-seminomatous testis cancer. Value of pre- and postorchiectomy serum tumor marker information in prediction of retroperitoneal lymph node metastases. Swedish-Norwegian Testicular Cancer Project (SWENOTECA). *Ann Oncol* **1**: 281-8

Klepp O, Olsson AM, Henrikson H, Aass N, Dahl O, Stenwig AE, Persson BE, Cavallin-Stahl E, Fossa SD, Wahlqvist L (1990b) Prognostic factors in clinical stage I nonseminomatous germ cell tumors of the testis: multivariate analysis of a prospective multicenter study. Swedish-Norwegian Testicular Cancer Group. *J Clin Oncol* **8:** 509-18

Krege S, Beyer J, Souchon R, Albers P, Albrecht W, Algaba F, Bamberg M, Bodrogi I, Bokemeyer C, Cavallin-Stahl E, Classen J, Clemm C, Cohn-Cedermark G, Culine S, Daugaard G, De Mulder PH, De Santis M, de Wit M, de Wit R, Derigs HG, Dieckmann KP, Dieing A, Droz JP, Fenner M, Fizazi K, Flechon A, Fossa SD, del Muro XG, Gauler T, Geczi L, Gerl A, Germa-Lluch JR, Gillessen S, Hartmann JT, Hartmann M, Heidenreich A, Hoeltl W, Horwich A, Huddart R, Jewett M, Joffe J, Jones WG, Kisbenedek L, Klepp O, Kliesch S, Koehrmann KU, Kollmannsberger C, Kuczyk M, Laguna P, Galvis OL, Loy V, Mason MD, Mead GM, Mueller R, Nichols C, Nicolai N, Oliver T, Ondrus D, Oosterhof GO, Ares LP, Pizzocaro G, Pont J, Pottek T, Powles T, Rick O, Rosti G, Salvioni R, Scheiderbauer J, Schmelz HU, Schmidberger H, Schmoll HJ, Schrader M, Sedlmayer F, Skakkebaek NE, Sohaib A, Tjulandin S, Warde P, Weinknecht S, Weissbach L, Wittekind C, Winter E, Wood L, von der Maase H (2008a) European consensus conference on diagnosis and treatment of germ cell cancer: a report of the second meeting of the European Germ Cell Cancer Consensus group (EGCCCG): part I. Eur Urol 53: 478-96

Krege S, Beyer J, Souchon R, Albers P, Albrecht W, Algaba F, Bamberg M, Bodrogi I, Bokemeyer C, Cavallin-Stahl E, Classen J, Clemm C, Cohn-Cedermark G, Culine S, Daugaard G, De Mulder PH, De Santis M, de Wit M, de Wit R, Derigs HG, Dieckmann KP, Dieing A, Droz JP, Fenner M, Fizazi K, Flechon A, Fossa SD, del Muro XG, Gauler T, Geczi L, Gerl A, Germa-Lluch JR, Gillessen S, Hartmann JT, Hartmann M, Heidenreich A, Hoeltl W, Horwich A, Huddart R, Jewett M, Joffe J, Jones WG, Kisbenedek L, Klepp O, Kliesch S, Koehrmann KU, Kollmannsberger C, Kuczyk M, Laguna P, Galvis OL, Loy V, Mason MD, Mead GM, Mueller R, Nichols C, Nicolai N, Oliver T, Ondrus D, Oosterhof GO, Paz-Ares L, Pizzocaro G, Pont J, Pottek T, Powles T, Rick O, Rosti G, Salvioni R, Scheiderbauer J, Schmelz HU, Schmidberger H, Schmoll HJ, Schrader M, Sedlmayer F, Skakkebaek NE, Sohaib A, Tjulandin S, Warde P, Weinknecht S, Weissbach L, Wittekind C, Winter E, Wood L, von der Maase H (2008b) European consensus conference on diagnosis and treatment of germ cell cancer: a report of the second meeting of the European Germ Cell Cancer Consensus Group (EGCCCG): part II. Eur Urol 53: 497-513

Kroenke K, Spitzer RL, Williams JB, Monahan PO, Lowe B (2007) Anxiety disorders in primary care: prevalence, impairment, comorbidity, and detection. *Ann Intern Med* **146**: 317-25

Kuczyk M, Machtens S, Bokemeyer C, Schultheiss D, Jonas U (2000) Sexual function and fertility after treatment of testicular cancer. *Curr Opin Urol* **10:** 473-7 Kupelian V, Page ST, Araujo AB, Travison TG, Bremner WJ, McKinlay JB (2006) Low sex hormone-binding globulin, total testosterone, and symptomatic androgen deficiency are associated with development of the metabolic syndrome in nonobese men. *J Clin Endocrinol Metab* **91:** 843-50

La Spada AR, Wilson EM, Lubahn DB, Harding AE, Fischbeck KH (1991) Androgen receptor gene mutations in X-linked spinal and bulbar muscular atrophy. *Nature* **352:** 77-9

Lackner J, Schatzl G, Koller A, Mazal P, Waldhoer T, Marberger M, Kratzik C (2005) Treatment of testicular cancer: influence on pituitarygonadal axis and sexual function. *Urology* **66:** 402-6

Lampe H, Horwich A, Norman A, Nicholls J, Dearnaley DP (1997) Fertility after chemotherapy for testicular germ cell cancers. *J Clin Oncol* **15**: 239-45

Lee DK, Chang C (2003) Molecular communication between androgen receptor and general transcription machinery. *J Steroid Biochem Mol Biol* **84:** 41-9

Lenz S, Giwercman A, Elsborg A, Cohr KH, Jelnes JE, Carlsen E, Skakkebaek NE (1993) Ultrasonic testicular texture and size in 444 men from the general population: correlation to semen quality. *Eur Urol* **24:** 231-8

Lenz S, Giwercman A, Skakkebaek NE, Bruun E, Frimodt-Moller C (1987) Ultrasound in detection of early neoplasia of the testis. *Int J Androl* **10:** 187-90

Lewis RW, Fugl-Meyer KS, Bosch R, Fugl-Meyer AR, Laumann EO, Lizza E, Martin-Morales A (2004) Epidemiology/risk factors of sexual dysfunction. *J Sex Med* 1: 35-9

Lowe B, Spitzer RL, Grafe K, Kroenke K, Quenter A, Zipfel S, Buchholz C, Witte S, Herzog W (2004) Comparative validity of three screening questionnaires for DSM-IV depressive disorders and physicians' diagnoses. *J Affect Disord* **78**: 131-40

Lundin KB, Giwercman A, Dizeyi N, Giwercman YL (2007) Functional in vitro characterisation of the androgen receptor GGN polymorphism. *Mol Cell Endocrinol* **264:** 184-7

Lundin KB, Giwercman A, Richthoff J, Abrahamsson PA, Giwercman YL (2003) No association between mutations in the human androgen receptor GGN repeat and inter-sex conditions. *Mol Hum Reprod* **9**: 375-9

Lundin KB, Giwercman YL, Rylander L, Hagmar L, Giwercman A (2006) Androgen receptor gene GGN repeat length and reproductive characteristics in young Swedish men. *Eur J Endocrinol* **155:** 347-54

Maddocks S, Hargreave TB, Reddie K, Fraser HM, Kerr JB, Sharpe RM (1993) Intratesticular hormone levels and the route of secretion of hormones from the testis of the rat, guinea pig, monkey and human. *Int J Androl* **16**: 272-8

Magelssen H, Haugen TB, von During V, Melve KK, Sandstad B, Fossa SD (2005) Twenty years experience with semen cryopreservation in testicular cancer patients: who needs it? *Eur Urol* **48**: 779-85

Matsumoto AM, Bremner WJ (1984) Modulation of pulsatile gonadotropin secretion by testosterone in man. *J Clin Endocrinol Metab* **58**: 609-14

McEwan IJ (2001) Structural and functional alterations in the androgen receptor in spinal bulbar muscular atrophy. *Biochem Soc Trans* **29**: 222-7

McLachlan RI, Matsumoto AM, Burger HG, de Kretser DM, Bremner WJ (1988) Relative roles of follicle-stimulating hormone and luteinizing hormone in the control of inhibin secretion in normal men. *J Clin Invest* **82:** 880-4

Meistrich ML (1993) Effects of chemotherapy and radiotherapy on spermatogenesis. *Eur Urol* **23:** 136-41; discussion 142

Mifsud A, Sim CK, Boettger-Tong H, Moreira S, Lamb DJ, Lipshultz LI, Yong EL (2001) Trinucleotide (CAG) repeat polymorphisms in the androgen receptor gene: molecular markers of risk for male infertility. *Fertil Steril* **75:** 275-81

Min JK, Williams KA, Okwuosa TM, Bell GW, Panutich MS, Ward RP (2006) Prediction of coronary heart disease by erectile dysfunction in men referred for nuclear stress testing. *Arch Intern Med* **166**: 201-6

Moller H (1998) Trends in sex-ratio, testicular cancer and male reproductive hazards: are they connected? *APMIS* **106**: 232-8; discussion 238-9

Moller H, Skakkebaek NE (1999) Risk of testicular cancer in subfertile men: case-control study. *BMJ* **318:** 559-62

Morishima A, Grumbach MM, Simpson ER, Fisher C, Qin K (1995) Aromatase deficiency in male and female siblings caused by a novel mutation and the physiological role of estrogens. *J Clin Endocrinol Metab* **80:** 3689-98

Murray CJ, Lopez AD (1997) Global mortality, disability, and the contribution of risk factors: Global Burden of Disease Study. *Lancet* **349**: 1436-42

Mykletun A, Dahl AA, Haaland CF, Bremnes R, Dahl O, Klepp O, Wist E, Fossa SD (2005) Side effects and cancer-related stress determine quality of life in long-term survivors of testicular cancer. *J Clin Oncol* **23:** 3061-8

Nazareth I, Lewin J, King M (2001) Sexual dysfunction after treatment for testicular cancer: a systematic review. *J Psychosom Res* **51**: 735-43

Nelson CJ, Nandy N, Roth AJ (2007) Chemotherapy and cognitive deficits: mechanisms, findings, and potential interventions. *Palliat Support Care* **5**: 273-80

Nichols CR, Breeden ES, Loehrer PJ, Williams SD, Einhorn LH (1993) Secondary leukemia associated with a conventional dose of etoposide: review of serial germ cell tumor protocols. *J Natl Cancer Inst* **85:** 36-40

Nieschlag E, Behre HM, Bouchard P, Corrales JJ, Jones TH, Stalla GK, Webb SM, Wu FC (2004) Testosterone replacement therapy: current trends and future directions. *Hum Reprod Update* **10**: 409-19

Nieschlag E, Behre HM, van Ahlen H (2001) *Andrology : male reproductive health and dysfunction*, 2. edn. Berlin: Springer

Nijman JM, Schraffordt Koops H, Kremer J, Sleijfer DT (1987) Gonadal function after surgery and chemotherapy in men with stage II and III nonseminomatous testicular tumors. *J Clin Oncol* **5:** 651-6

Nord C, Bjoro T, Ellingsen D, Mykletun A, Dahl O, Klepp O, Bremnes RM, Wist E, Fossa SD (2003) Gonadal hormones in long-term survivors 10 years after treatment for unilateral testicular cancer. *Eur Urol* **44:** 322-8

Nuver J, Smit AJ, van der Meer J, van den Berg MP, van der Graaf WT, Meinardi MT, Sleijfer DT, Hoekstra HJ, van Gessel AI, van Roon AM, Gietema JA (2005a) Acute chemotherapy-induced cardiovascular changes in patients with testicular cancer. *J Clin Oncol* **23**: 9130-7

Nuver J, Smit AJ, Wolffenbuttel BH, Sluiter WJ, Hoekstra HJ, Sleijfer DT, Gietema JA (2005b) The metabolic syndrome and disturbances in hormone levels in long-term survivors of disseminated testicular cancer. *J Clin Oncol* **23**: 3718-25

Ochsenkuhn R, Kamischke A, Nieschlag E (1999) Imipramine for successful treatment of retrograde ejaculation caused by retroperitoneal surgery. *Int J Androl* **22:** 173-7

Ohl DA, Denil J, Bennett CJ, Randolph JF, Menge AC, McCabe M (1991) Electroejaculation following retroperitoneal lymphadenectomy. *J Urol* **145**: 980-3

Oliver RT, Mason MD, Mead GM, von der Maase H, Rustin GJ, Joffe JK, de Wit R, Aass N, Graham JD, Coleman R, Kirk SJ, Stenning SP (2005) Radiotherapy versus single-dose carboplatin in adjuvant treatment of stage I seminoma: a randomised trial. *Lancet* **366**: 293-300

Osterlind A, Berthelsen JG, Abildgaard N, Hansen SO, Hjalgrim H, Johansen B, Munck-Hansen J, Rasmussen LH (1991) Risk of bilateral testicular germ cell cancer in Denmark: 1960-1984. *J Natl Cancer Inst* 83: 1391-5

Palmieri G, Lotrecchiano G, Ricci G, Spiezia R, Lombardi G, Bianco AR, Torino G (1996) Gonadal function after multimodality treatment in men with testicular germ cell cancer. *Eur J Endocrinol* **134**: 431-6

Paulozzi LJ, Erickson JD, Jackson RJ (1997) Hypospadias trends in two US surveillance systems. *Pediatrics* **100**: 831-4

Pedersen-Bjergaard J, Daugaard G, Hansen SW, Philip P, Larsen SO, Rorth M (1991) Increased risk of myelodysplasia and leukaemia after etoposide, cisplatin, and bleomycin for germ-cell tumours. *Lancet* **338:** 359-63

Pedersen AD, Rossen P, Mehlsen MY, Pedersen CG, Zachariae R, H VDM (2009) Long-term cognitive function following chemotherapy in patients with testicular cancer. *J Int Neuropsychol Soc*: 1-6
Petersen PM, Daugaard G, Rorth M, Skakkebaek NE (2003) Endocrine function in patients treated for carcinoma in situ in the testis with irradiation. *APMIS* **111:** 93-8; discussion 98-9

Petersen PM, Giwercman A, Daugaard G, Rorth M, Petersen JH, Skakkeaek NE, Hansen SW, von der Maase H (2002) Effect of graded testicular doses of radiotherapy in patients treated for carcinoma-in-situ in the testis. *J Clin Oncol* **20**: 1537-43

Petersen PM, Hansen SW, Giwercman A, Rorth M, Skakkebaek NE (1994) Dose-dependent impairment of testicular function in patients treated with cisplatin-based chemotherapy for germ cell cancer. *Ann Oncol* **5**: 355-8

Petersen PM, Skakkebaek NE, Rorth M, Giwercman A (1999a) Semen quality and reproductive hormones before and after orchiectomy in men with testicular cancer. *J Urol* **161:** 822-6

Petersen PM, Skakkebaek NE, Vistisen K, Rorth M, Giwercman A (1999b) Semen quality and reproductive hormones before orchiectomy in men with testicular cancer. *J Clin Oncol* **17:** 941-7

Pont J, Albrecht W (1997) Fertility after chemotherapy for testicular germ cell cancer. *Fertil Steril* **68:** 1-5

Rajpert-De Meyts E, Leffers H, Petersen JH, Andersen AG, Carlsen E, Jorgensen N, Skakkebaek NE (2002) CAG repeat length in androgenreceptor gene and reproductive variables in fertile and infertile men. *Lancet* **359:** 44-6

Raman JD, Nobert CF, Goldstein M (2005) Increased incidence of testicular cancer in men presenting with infertility and abnormal semen analysis. *J Urol* **174:** 1819-22; discussion 1822

Robinson D, Moller H, Horwich A (2007) Mortality and incidence of second cancers following treatment for testicular cancer. *Br J Cancer* **96**: 529-33

Rosenlund B, Sjoblom P, Tornblom M, Hultling C, Hillensjo T (1998) Invitro fertilization and intracytoplasmic sperm injection in the treatment of infertility after testicular cancer. *Hum Reprod* **13**: 414-8

Rowley MJ, Leach DR, Warner GA, Heller CG (1974) Effect of graded doses of ionizing radiation on the human testis. *Radiat Res* **59**: 665-78

Sedlmayer F, Joos H, Deutschmann H, Rahim H, Merz F, Kogelnik HD (1999) [Long-term tumor control and fertility after para-aortic limited radiotherapy of stage I seminoma]. *Strahlenther Onkol* **175**: 320-4

Selby C (1990) Sex hormone binding globulin: origin, function and clinical significance. *Ann Clin Biochem* **27** (**Pt 6**): 532-41

Sheckter CB, Matsumoto AM, Bremner WJ (1989) Testosterone administration inhibits gonadotropin secretion by an effect directly on the human pituitary. *J Clin Endocrinol Metab* **68:** 397-401

Skakkebaek NE, Rajpert-De Meyts E, Main KM (2001) Testicular dysgenesis syndrome: an increasingly common developmental disorder with environmental aspects. *Hum Reprod* **16**: 972-8

Skoogh J, Steineck, G., Stierner, U. K., Cavallin-Ståhl, E., Olofsson, U., Wallin, A., Gatz, M., Johansson, B. (2008) Long-term cognitive function among testicular cancer survivors treated with chemotherapy. *Abstarct, ASCO meeting, J Clin Oncol 26: 2008 (May 20 suppl; abstr 5035)*

Sofikitis N, Giotitsas N, Tsounapi P, Baltogiannis D, Giannakis D, Pardalidis N (2008) Hormonal regulation of spermatogenesis and spermiogenesis. *J Steroid Biochem Mol Biol* **109:** 323-30

Spano M, Bonde JP, Hjollund HI, Kolstad HA, Cordelli E, Leter G (2000) Sperm chromatin damage impairs human fertility. The Danish First Pregnancy Planner Study Team. *Fertil Steril* **73:** 43-50

Spector IP, Carey MP (1990) Incidence and prevalence of the sexual dysfunctions: a critical review of the empirical literature. *Arch Sex Behav* **19:** 389-408

Stahl O, Eberhard J, Jepson K, Spano M, Cwikiel M, Cavallin-Stahl E, Giwercman A (2004) The impact of testicular carcinoma and its treatment on sperm DNA integrity. *Cancer* **100**: 1137-44

Stahl O, Eberhard J, Jepson K, Spano M, Cwikiel M, Cavallin-Stahl E, Giwercman A (2006) Sperm DNA integrity in testicular cancer patients. *Hum Reprod* **21:** 3199-205

Stanworth RD, Kapoor D, Channer KS, Jones TH (2008) Androgen receptor CAG repeat polymorphism is associated with serum testosterone levels, obesity and serum leptin in men with type 2 diabetes. *Eur J Endocrinol* **159**: 739-46

Stephenson WT, Poirier SM, Rubin L, Einhorn LH (1995) Evaluation of reproductive capacity in germ cell tumor patients following treatment with cisplatin, etoposide, and bleomycin. *J Clin Oncol* **13**: 2278-80

Strumberg D, Brugge S, Korn MW, Koeppen S, Ranft J, Scheiber G, Reiners C, Mockel C, Seeber S, Scheulen ME (2002) Evaluation of longterm toxicity in patients after cisplatin-based chemotherapy for nonseminomatous testicular cancer. *Ann Oncol* **13**: 229-36

Su QR, Su LY, Su HR, Chen Q, Ren GY, Yin Y, Shen SQ, Yu AY, Xia GY (2007) Polymorphisms of androgen receptor gene in childhood and adolescent males with first-onset major depressive disorder and association with related symptomatology. *Int J Neurosci* **117**: 903-17

Swan SH, Elkin EP, Fenster L (1997) Have sperm densities declined? A reanalysis of global trend data. *Environ Health Perspect* **105**: 1228-32

Swerdlow AJ, Higgins CD, Pike MC (1997) Risk of testicular cancer in cohort of boys with cryptorchidism. *BMJ* **314:** 1507-11

Tapanainen JS, Vaskivuo T, Aittomaki K, Huhtaniemi IT (1998) Inactivating FSH receptor mutations and gonadal dysfunction. *Mol Cell Endocrinol* **145**: 129-35

Tchernof A, Despres JP (2000) Sex steroid hormones, sex hormone-binding globulin, and obesity in men and women. *Horm Metab Res* **32:** 526-36

Thompson IM, Tangen CM, Goodman PJ, Probstfield JL, Moinpour CM, Coltman CA (2005) Erectile dysfunction and subsequent cardiovascular disease. *JAMA* **294:** 2996-3002

Tinkler SD, Howard GC, Kerr GR (1992) Sexual morbidity following radiotherapy for germ cell tumours of the testis. *Radiother Oncol* **25:** 207-12

Travis LB, Fossa SD, Schonfeld SJ, McMaster ML, Lynch CF, Storm H, Hall P, Holowaty E, Andersen A, Pukkala E, Andersson M, Kaijser M, Gospodarowicz M, Joensuu T, Cohen RJ, Boice JD, Jr., Dores GM, Gilbert ES (2005) Second cancers among 40,576 testicular cancer patients: focus on long-term survivors. *J Natl Cancer Inst* **97:** 1354-65

Tut TG, Ghadessy FJ, Trifiro MA, Pinsky L, Yong EL (1997) Long polyglutamine tracts in the androgen receptor are associated with reduced trans-activation, impaired sperm production, and male infertility. *J Clin Endocrinol Metab* **82:** 3777-82

Wallace AM, Tucker P, Williams DM, Hughes IA, Ahmed SF (2003) Shortterm effects of prednisolone and dexamethasone on circulating concentrations of leptin and sex hormone-binding globulin in children being treated for acute lymphoblastic leukaemia. *Clin Endocrinol (Oxf)* **58**: 770-6

Vanderschueren D, Boonen S, Ederveen AG, de Coster R, Van Herck E, Moermans K, Vandenput L, Verstuyf A, Bouillon R (2000) Skeletal effects of estrogen deficiency as induced by an aromatase inhibitor in an aged male rat model. *Bone* **27**: 611-7

Vaughn DJ, Palmer SC, Carver JR, Jacobs LA, Mohler ER (2008) Cardiovascular risk in long-term survivors of testicular cancer. *Cancer* **112:** 1949-53

Veldhuis JD, Rogol AD, Johnson ML (1983) Endogenous opiates modulate the pulsatile secretion of biologically active luteinizing hormone in man. *J Clin Invest* **72:** 2031-40

Verdecchia A, Francisci S, Brenner H, Gatta G, Micheli A, Mangone L, Kunkler I (2007) Recent cancer survival in Europe: a 2000-02 period analysis of EUROCARE-4 data. *Lancet Oncol* **8**: 784-96

Wethal T, Kjekshus J, Roislien J, Ueland T, Andreassen AK, Wergeland R, Aukrust P, Fossa SD (2007) Treatment-related differences in cardiovascular risk factors in long-term survivors of testicular cancer. *J Cancer Surviv* **1**: 8-16

WHO Task Force on the Diagnosis and Treatment of Infertility., Comhaire FH, World Health Organization. (1987) *Towards more objectivity in diagnosis and management of male infertility : results of a World Health Organization multicentre study*. Oxford: Blackwell Scientific

Wiechno P, Demkow T, Kubiak K, Sadowska M, Kaminska J (2007) The quality of life and hormonal disturbances in testicular cancer survivors in Cisplatin era. *Eur Urol* **52:** 1448-54

Willemse PH, Sleijfer DT, Sluiter WJ, Schraffordt Koops H, Doorenbos H (1983) Altered Leydig cell function in patients with testicular cancer: evidence for bilateral testicular defect. *Acta Endocrinol (Copenh)* **102:** 616-24

Winters SJ (1991) Diurnal rhythm of testosterone and luteinizing hormone in hypogonadal men. *J Androl* **12:** 185-90

Winters SJ, Troen P (1986) Testosterone and estradiol are co-secreted episodically by the human testis. *J Clin Invest* **78**: 870-3

Wittchen HU, Hofler M, Meister W (2001) Prevalence and recognition of depressive syndromes in German primary care settings: poorly recognized and treated? *Int Clin Psychopharmacol* **16**: 121-35

Wittchen HU, Kessler RC, Beesdo K, Krause P, Hofler M, Hoyer J (2002) Generalized anxiety and depression in primary care: prevalence, recognition, and management. *J Clin Psychiatry* **63 Suppl 8:** 24-34

von Eckardstein S, Syska A, Gromoll J, Kamischke A, Simoni M, Nieschlag E (2001a) Inverse correlation between sperm concentration and number of androgen receptor CAG repeats in normal men. *J Clin Endocrinol Metab* **86:** 2585-90

von Eckardstein S, Tsakmakidis G, Kamischke A, Rolf C, Nieschlag E (2001b) Sonographic testicular microlithiasis as an indicator of premalignant conditions in normal and infertile men. *J Androl* **22**: 818-24

Wood HM, Elder JS (2009) Cryptorchidism and testicular cancer: separating fact from fiction. *J Urol* **181:** 452-61

World Health Organization. (1999) *WHO laboratory manual for the examination of human semen and sperm-cervical mucus interaction*, 4th edn. Cambridge: Published on behalf of the World Health Organization by Cambridge University Press

WorldHealthOrganization (2000) WHO Manual for the standardised Investigation, Diagnosis and Management of the Infertile Man. Yaffe K, Edwards ER, Lui LY, Zmuda JM, Ferrell RE, Cauley JA (2003) Androgen receptor CAG repeat polymorphism is associated with cognitive function in older men. *Biol Psychiatry* **54:** 943-6

Zhang FP, Pakarainen T, Poutanen M, Toppari J, Huhtaniemi I (2003) The low gonadotropin-independent constitutive production of testicular testosterone is sufficient to maintain spermatogenesis. *Proc Natl Acad Sci U S A* **100:** 13692-7

Zigmond AS, Snaith RP (1983) The hospital anxiety and depression scale. *Acta Psychiatr Scand* **67:** 361-70

Original publications